- 11.3.4 a significant change to any risk component;
- 11.3.5 a significant change in the quantity of Personal Data held within the Information Management System;
- 11.3.6 a proposal to change any of the Sites from which any part of the Deliverables are provided;
- 11.3.7 a change in any Subcontractor involved in the provision of the Deliverables; or
- 11.3.8 an ISO 27001 audit report produced in connection with the Certification Requirements indicating significant concerns.
- 11.4 Any identified risks, vulnerabilities, or other security concerns that are rated as Critical shall be notified as soon as possible, and within one hour. Notification to include email, telephone and other measures, and the supplier must secure acknowledgement before considering this SLA to be met. For High, this period may be extended to 1 working day, and for all other topics the period is 2 Working Days.
- 11.5 Within 10 Working Days of such notification to the Buyer or such other timescale as may be agreed with the Buyer, the Supplier shall make the necessary changes to the Required Changes Register and submit the updated Required Changes Register to the Buyer for review and approval. Depending on the impact of the risks being mitigated, this timescale may be considerably shorter and the buyer reserves the right to require priority and / or emergency changes for remediation of Critical and High severity risks.
- 11.6 Where the Supplier is required to implement a change, including any change to the Information Management System, in order to remedy any non-compliance with this Contract, the Supplier shall effect such change at its own cost and expense and within the timescales set out in the Required Changes Register.
- 11.7 The Buyer may require, and the Supplier shall provide the Buyer and its authorised representatives with:
  - 11.7.1 access to the Supplier Staff;
  - 11.7.2 access to the Information Management System to audit the Supplier and its Subcontractors' compliance with this Contract; and
  - 11.7.3 such other information and/or documentation that the Buyer or its authorised representatives may reasonably require;
  - 11.7.4 to assist the Buyer to establish whether the arrangements which the Supplier and its Subcontractors have implemented in order to ensure the security of the Government Data and the Information Management System are consistent with the representations in the Information Security Management Document Set. The Supplier shall provide the access required by the Buyer in accordance with

Call-Off Schedule 9A (Health Security) Call-Off Ref: C52340 PD - NHS Pathways Crown Copyright 2018

> this Paragraph within 10 Working Days of receipt of such request, except in the case of a Breach of Security in which case the Supplier shall provide the Buyer with the access that it requires within 24 hours of receipt of such request.

### 12 Security Testing

- 12.1 The Supplier shall, at its own cost and expense procure and conduct Security Testing of the Supplier's system, including any subcontracted systems used to provide the Deliverables to the Buyer. If specified by the Buyer this must be undertaken by a CHECK Service Provider, otherwise this may be undertaken by a service provider under the CREST, TIGER or Cyber scheme:
  - 12.1.1 Crest https://www.crest-approved.org/
  - 12.1.2 Tiger https://www.tigerscheme.org/.
- 12.2 All Security Testing must be scoped so as to provide a realistic assessment of the efficacy of the Supplier's ISMS and risk mitigations. Service providers carrying out Security Testing should be provided with the security designs that they are validating and reports should include an assessment as to whether the existing control set is in line with the expected mitigations.
- 12.3 The Supplier shall complete all of the Security Testing before the Supplier submits the Security Assurance Statement to the Buyer for review in accordance with Paragraph 5 of this Annex, and repeat the Security Testing not less than once every 12 months and upon any significant change to the Supplier's system during the Call-Off Contract Period and submit the results of each such test to the Buyer for review in accordance with this Paragraph.
- 12.4 Reports and results of the Security Testing shall be made available for Buyer review such that the Buyer can have confidence and assurance over the residual risk of the Supplier's system.
- 12.5 If Security Testing is required to be carried out by a CHECK Service Provider pursuant to Paragraph 12.1 the Supplier shall:
  - 12.5.1 agree with the Buyer the aim and scope of the relevant Security Testing; and
  - 12.5.2 promptly, following receipt of each Security Testing report, provide the Buyer with a copy of the report.
- 12.6 in the event that the Security Testing report identifies any vulnerabilities, the Supplier shall prepare a remedial plan for approval by the Buyer (each a "Vulnerability Correction Plan") which sets out in respect of each vulnerability identified in the Security Testing report:
  - 12.6.1 how the vulnerability will be remedied;

- 12.6.2 the date by which the vulnerability will be remedied; and
- 12.6.3 the tests which the Supplier shall perform or procure to be performed (which may, at the discretion of the Buyer, include further Security Testing) to confirm that the vulnerability has been remedied.
- 12.7 The Supplier shall comply with the Vulnerability Correction Plan and conduct such further tests on the Supplier's system as are required by the Vulnerability Correction Plan to confirm that the Vulnerability Correction Plan has been complied with.
- 12.8 The Supplier shall ensure that any Security Testing which could adversely affect the Supplier's system shall be designed and implemented by the Supplier so as to minimise the impact, on the delivery of the Deliverables, for example by using a representative test environment, and the date, timing, content and conduct of such tests shall be agreed in advance with the Buyer.
- 12.9 If any testing conducted by or on behalf of the Supplier identifies a new risk new threat, vulnerability or exploitation technique that has the potential to affect the security of the Information Management System, the Supplier shall within 2 days of becoming aware of such risk, threat, vulnerability or exploitation technique provide the Buyer with a copy of the unredacted test report and:
  - 12.9.1 propose interim mitigation measures to vulnerabilities in the Information Management System known to be exploitable where a security patch is not immediately available; and
  - 12.9.2 where and to the extent applicable, remove or disable any extraneous interfaces, services or capabilities that are not needed for the provision of the Deliverables (in order to reduce the attack surface of the Supplier's system) within the timescales set out in the test report or such other timescales as may be agreed with the Buyer.
- 12.10 The Supplier shall conduct such further tests of the Supplier's system as may be required by the Buyer from time to time to demonstrate compliance with its obligations set out this Schedule and the Contract.
- 12.11 The Supplier shall notify the Buyer immediately if it fails to, or believes that it will not, mitigate the vulnerability within the timescales set out in this Annex.

### 13 Security Monitoring and Reporting

- 13.1 The Supplier shall:
  - 13.1.1 monitor the delivery of assurance activities;
  - 13.1.2 maintain and update the Security Assurance Statement in accordance with Paragraph 5 of this Annex;

- 13.1.3 monitor security risks impacting upon the operation of the Deliverables;
- 13.1.4 monitor the Information Management System for attempted Breaches of Security, including but not limited to, failed authentication, attempted brute force, indications of attempted denial of service attacks, attempted or actual data exfiltration, suspicious system alterations, and privileged access;
- 13.1.5 report actual or attempted Breaches of Security in accordance with the approved Incident Management Process; and
- 13.1.6 agree with the Buyer the frequency and nature of the security reports to be prepared and submitted by the Supplier to the Buyer within 30 days of the Start Date of this Call-Off Contract.

### 14 Vulnerabilities and Corrective Action

- 14.1 The Buyer and the Supplier acknowledge that from time to time vulnerabilities in the Information Management System will be discovered which unless mitigated will present an unacceptable risk to the Government Data.
- 14.2 The severity of vulnerabilities for Supplier COTS Software and Third Party COTS Software shall be categorised by the supplier as 'critical', 'important' and 'other' by aligning these categories to the vulnerability scoring according to the agreed method in the Information Security Management Document Set and using the appropriate vulnerability scoring systems including:
  - 14.2.1 the 'National Vulnerability Database' 'Vulnerability Severity Ratings': 'High', 'Medium' and 'Low' respectively (these in turn are aligned to CVSS scores as set out by NIST at http://nvd.nist.gov/cvss.cfm); and
  - 14.2.2 Microsoft's 'Security Bulletin Severity Rating System' ratings 'Critical', 'Important', and the two remaining levels ('Moderate' and 'Low') respectively.
- 14.3 Subject to Paragraphs 14.4 and 14.5 and of this Annex, the Supplier shall procure the application of security patches to vulnerabilities in the core Information Management System within:
  - 14.3.1 2 days after the public release of patches for those vulnerabilities categorised as 'critical';
  - 14.3.2 30 days after the public release of patches for those vulnerabilities categorised as 'important'; and
  - 14.3.3 60 days after the public release of patches for those vulnerabilities categorised as 'other'.
- 14.4 Where a vulnerability is discovered or reasonably suspected to be under active

exploitation upon discovery, or within the agreed remediation timeframe, and posing an active risk to Government Data, the timeframes set out in Paragraph 14.3 shall cease to apply and the remediation will be escalated as an emergency and progressed as soon as possibly in active consultation with the Buyer.

- 14.5 The timescales for applying patches to vulnerabilities in the core Information Management System set out in Paragraph 14.3 of this Annex shall be extended (subject to Buyer agreement) where:
  - 14.5.1 the Supplier can demonstrate that a vulnerability in the core Information Management System is not exploitable within the context of the Deliverables (e.g. because it resides in a software component which is not involved in running in the Deliverables) provided such vulnerabilities shall be remedied by the Supplier within the timescales set out in Paragraph 14.3 of this Annex if the vulnerability becomes exploitable within the context of the Deliverables;
  - 14.5.2 the application of a 'Critical' or 'Important' security patch adversely affects the Supplier's ability to deliver the Deliverables in which case the Supplier shall be granted an extension to such timescales of 5 days, provided the Supplier had followed and continues to follow the security patch test plan agreed with the Buyer; or
  - 14.5.3 the Buyer agrees a different maximum period after a case-by-case consultation with the Supplier under the processes defined in the Information Security Management Document Set.
- 14.6 The Information Security Management Document Set shall include provisions for major version upgrades of all Supplier COTS Software and Third Party COTS Software to be kept up to date such that all Supplier COTS Software and Third Party COTS Software are always in mainstream support throughout the Call-Off Contract Period unless otherwise agreed by the Buyer in writing.

### 15 Breach of Security

- 15.1 If either Party becomes aware of an actual or attempted Breach of Security, it shall notify the other in accordance with the Incident Management Process.
- 15.2 The Incident Management Process shall, as a minimum, require the Supplier to do the following upon it becoming aware of a Breach of Security or attempted Breach of Security:
  - 15.2.1 immediately take all reasonable steps necessary to:
    - minimise the extent of actual or potential harm caused by such Breach of Security;
    - remedy such Breach of Security to the extent possible;

Call-Off Schedule 9A (Health Security) Call-Off Ref: C52340 PD - NHS Pathways Crown Copyright 2018

- apply a tested mitigation against any such Breach of Security;
- prevent a further Breach of Security in the future which exploits the same root cause failure; and
- preserve any evidence that may be relevant to any internal, Buyer or regulatory investigation or criminal or legal proceedings;
- 15.2.2 notify the Buyer immediately upon becoming aware of a Breach of Security or attempted Breach of Security or circumstances that are likely to give rise to a Breach of Security, providing the Buyer with sufficient information to meet any obligations to report a Breach of Security involving any Personal Data under the Data Protection Legislation; and
- 15.2.3 as soon as reasonably practicable and, in any event, within 2 Working Days, following the Supplier becoming aware of the Breach of Security or attempted Breach of Security, provide to the Buyer full details of the Breach of Security or attempted Breach of Security, including a root cause analysis where required by the Buyer.
- 15.3 In the event that any action is taken in response to a Breach of Security or attempted Breach of Security as a result of non-compliance by the Supplier, its Subcontractors and/or all or any part of the Information Management System with this Contract, then such remedial action shall be completed at no additional cost to the Buyer.

### 16 Termination Rights

- 16.1 Without limitation, the following events shall constitute a material Default giving the Buyer a right to terminate for cause pursuant to Clause 10.4.1(d) of the Core Terms:
  - 16.1.1 the Buyer issues two rejection notices in respect of the Security Assurance Statement;
  - 16.1.2 the Supplier fails to implement a change required by the Required Changes Register in accordance with the timescales set out in the Required Changes Register;
  - 16.1.3 the Supplier fails to patch vulnerabilities in accordance with Paragraph 14 of Annex 3;
  - 16.1.4 the Supplier materially fails to comply with the Incident Management Process;
  - 16.1.5 the Supplier fails to meet the Certification Requirements;
  - 16.1.6 the Supplier fails to comply with any Vulnerability Correction Plan; or

- 16.1.7 the Supplier experiences an event analogous to a Breach of Security in respect of its own or any other customers' data and any contributing factor to such event:
  - would be a cause for termination pursuant to this Paragraph 16 had such event been a Breach of Security pursuant to this Contract; or
  - b) demonstrates a failure to meet the requirements of this Schedule that gives the Buyer a right to terminate pursuant to this Paragraph 16.

Call-Off Schedule 9A (Health Security) Call-Off Ref: C52340 PD - NHS Pathways Crown Copyright 2018

### Annex 4

### Information Security Management Document Set Template

	The Information Security Management Document Set Template is required to be completed		
. 1		200-2	

The template may be found as a stand-alone file associated with this schedule.

### Call-Off Schedule 10A (Health Exit Management)

### 1 Definitions

1.1 In this Schedule, the following words shall have the following meanings andthey shall supplement Joint Schedule 1 (Definitions):

"Exit Plan"	means the Exit Plan to be agreed by the parties in
	accordance with the provisions of Call-Off Schedule
	10A; and

<sup>&</sup>quot;Final Exit Plan" has the meaning given to it in paragraph 4.1 of Call-Off Schedule 10A.

### 2 Handovers between Statements of Work

- 2.1 Every Statement of Work must include, as part of its final activities, provisions for handover to any subsequent and dependent Statement of Works.
- 2.2 Handovers should include any necessary documentation, training, and data necessary to allow for successful transition or exit, should the latter be decided upon.

### 3 Exit Plan

### 3.1 Introduction

- 3.1.1 Within 2 months of the Start Date (or as otherwise agreed between the Buyer and Supplier), the Supplier shall prepare a draft Exit Plan in accordance with Good Industry Practice and the provisions set out below, and shall provide such draft Exit Plan to the Buyer to review and approve.
- 3.1.2 The Buyer and the Supplier shall together review the draft Exit Plan, and shall aim to agree the draft Exit Plan within 3 months of the Start Date.
- 3.1.3 The Supplier shall at any time during the Call-Off Contract Period provide an updated draft Exit Plan where the provision of the Deliverables materially changes and this impacts the provisions of the Exit Plan.
- 3.1.4 The Parties shall annually jointly review, and the Supplier shall update if necessary, the provisions of the Exit Plan.

### 3.2 Content of Plan

3.2.1 The Supplier shall ensure that the Exit Plan facilitates a Service Transfer to the Buyer or a Replacement Supplier on expiry or termination of the Call Off Contract.

- 3.2.2 As a minimum the Exit Plan will include:
  - Provision of / access to key Service information, workbook data, Supplier data, key Buyer processes and requirements, and TUPE information;
  - Management structure throughout the exit;
  - Roles and Responsibilities, which may include:

Role	Responsibilities
Exit Manager	Management of all Workstreams, including Communications and Finance
Project Management Support	Support across all Workstreams
Framework Director	Project Governance
Data Lead	Data & Reporting Workstreams
Technology Lead	Technology Workstream
TUPE lead	People Workstream
Supplier Lead	Supplier Management Workstream
Operations and Delivery Lead	Operations & WIP Workstreams

- Activities and timeline for the exit The exact nature of the activities and the timelines associated with them will be dependent on the planning and activities defined by the Buyer and the Replacement Supplier, most notably the timelines and phasing of the specific Buyer roll outs, and the associated implications. The Exit Plan should nevertheless incorporate indicative timescales and milestones with these to be firmed up by agreement between the Parties no later than an agreed timespan before the date of expiry or termination;
- Logical workstreams into which the activities will be organised, which may, for example, include:

### Call-Off Schedule 10A (Health Exit Management) Call-Off Ref: C52340 PD - NHS Pathways

Call-Off Ref: C52340 PD - NHS Pathway Crown Copyright 2018

Workstream	Key Activities
	Identify Exit Manager
	Identify Data Lead
	Identify Exit Board and key sponsors
	Understand scope and scale of new
	service, phasing, etc
Project Governance	Confirm exit activities and timelines
	Establish and maintain RAID Log
	Align exit activities to agreed exit timelines
	Sign off detailed plan and activities
	Identify Project Workstream contributors
	Agree timeline to control closure of access to any Buyer Systems
Technology	ALS SAL
	The Supplier to review data requests and provide workforce data in
	reasonable format and frequency.
	Supplier to provide a final data cut during hold/freeze period in line with WIP requirements
Data	Agree data archiving approach and data deletion as required by the Buyer, in line with GDPR & contractual requirements.
	Agree how data will be transferred a exit, including encryption
	Buyer data requirements to be finalised re retirement of incumbent workflow
	Provide Buyer specific process maps and variations
Operations & Delivery	Provide responses to reasonable Replacement Supplier clarification requests
People	Provide a point of contact in HR to agree TUPE timelines & approach

### Call-Off Schedule 10A (Health Exit Management)

Call-Off Ref: C52340 PD - NHS Pathways Crown Copyright 2018

Workstream	Key Activities
	Activities as required to comply with Part E of Call-Off Schedule 2 (Staff Transfer)
	Provide all current suppliers and contact details
Supplier Management	Support reasonable communications to suppliers and issue any required communications
	Manage billing closure with Supplier
	Feed into communications plan
	Feed into communications drafting
	Ensure all relevant Supplier teams understand activities/ progress of exit / agreed messaging
Communications and Change Management	Fully brief helpdesk on FAQs and messaging
	Issue communications to workers and suppliers as per plan
	Provide input to change impact assessment
Reporting	Provide a detailed overview of current reporting suite detailing key criteria, recipients and frequency
~///	Agree process & commercial arrangements for WIP transition
Work in Progress Transition (WIP)	Support data cleanse activity with a final data cut submitted to incoming service provider
	Support WIP freeze on raising new requisitions and worker changes
Finance	Provide final billing and confirm final time sheeting details
	Support in closing down purchase orders (if applicable)
	Support communication to workers and suppliers on billing transition

 Details of the transition of Deliverables, processes, data etc during the exit;

- Details of how technologies and accesses will be retired;
- Issue management governance structure; and
- Key assumptions, which may, for example, include;
  - Data Requests to be reasonable, specific and where necessary have clear articulation of why such data is required;
  - Response Timelines timelines for activities and data requests to be reasonable and reflect the work effort required in producing / executing;
  - Active Engagement –Supplier to be kept fully informed of Buyer progress and updates; and
  - Buyer Points of Contact –provide dedicated resource to support in the management of the exit and help manage issues and escalations.

### 4 Exit Management

- 4.1 The Supplier agrees that within 20 Working Days of the earliest of:
  - 4.1.1 receipt of a notification from the Buyer of a Service Transfer or intended Service Transfer;
  - 4.1.2 receipt of the giving of notice of early termination or any Partial Termination of the relevant Contract;
  - 4.1.3 the date which is 12 Months before the end of the Term; and
  - 4.1.4 receipt of a written request of the Buyer at any time,

the Supplier shall provide a complete set of information it is required to provide under the Exit Plan and the Parties shall agree the dates for completion of the activities set out in the Exit Plan. The Exit Plan, once populated with dates for the completion of activities ("Final Exit Plan") shall govern exit and transition of the Deliverables.

- 4.2 In relation to the delivery of the activities in a Final Exit Plan for a Service Transfer, the Supplier shall provide all reasonable co-operation and collaboration with the Buyer and Replacement Supplier including to agree aligned dates and to perform, and facilitate the performance of, aligned activities.
- 4.3 To the extent it does not adversely affect the Supplier's performance of any remaining Deliverables, then for the purposes of executing a Final Exit Plan, the Supplier shall:
  - 4.3.1 cease to use the Government Data (subject to paragraph 4.5);
  - 4.3.2 comply with the deletion requirements described in paragraph 4.4

as impacted by paragraph 4.5;

- 4.3.3 return to the Buyer all of the following if it is in the Supplier's possession or control:
  - all copies of Buyer Software licensed or provided by the Buyer;
  - all materials and documents owned by the Buyer; and
  - any other Buyer Assets provided by the Buyer.
- 4.4 Subject to paragraph 4.5, the Supplier shall as soon as reasonably practicable after termination of the Deliverables return (if required by the Buyer) all Government Data and any copies of it or of the information it contains, and in any case securely and irrevocably delete from its systems the Government Data in accordance with the applicable provisions of Call Off Schedule 9A (Health Security). The Supplier shall certify that all copies of the Government Data have been deleted within a reasonable time and in any event not later than 90 days after termination of the Deliverables.
- 4.5 The Supplier may continue to Process Personal Data contained within the Government Data following termination of the Deliverables to the extent necessary to support access by the Controllers to historical activity or audit data contained in the Supplier's systems where set out as required and in accordance with the conditions set out in Joint Schedule 11 (Processing Data).
- 4.6 When the Supplier believes that it has completed all activities in a Final Exit Plan, the Supplier shall notify the Buyer who shall then assess whether it is satisfied that the activities have been successfully completed. If the Buyer agrees that the Supplier has completed all of the required activities for that particular Final Exit Plan, it shall confirm its agreement in writing. If the Buyer does not agree with the Supplier's assertion that it has completed all of the required activities, then it shall notify the Supplier of the reasons why and following receipt of such reasons, the Supplier shall complete the required outstanding actions in a timeframe as will be reasonably agreed between the Parties.

### 5 Confidential Information

5.1 Subject to the requirements of Joint Schedule 11 (Processing Data) in relation to data retention, return and destruction, upon termination or expiry of this Call Off Contract, each Party shall return to the other Party (or if requested, destroy or delete) all Confidential Information of the other Party and shall certify that it does not retain the other Party's Confidential Information save to the extent (and for the limited period) that such information needs to be retained by the Party in question for the purposes of completing a Service Transfer or for statutory compliance purposes. The parties agree that any

Personal Data will be managed in accordance with Joint Schedule 11 (Processing Data).

5.2 The Supplier agrees that any Final Exit Plan agreed pursuant to the process described in paragraph 4.1 may be shared with CCS and with the Replacement Supplier(s).

### 6 Charges

6.1 Each Party shall bear its own costs in relation to the performance of its obligations described in this schedule.

### Call-Off Schedule 13A (Health Implementation Plan and Testing)

### Part A - Implementation

### 1. DEFINITIONS

1.1 In this Schedule, the following words shall have the following meanings and they shall supplement Joint Schedule 1 (Definitions):

"Delay"

"Deliverable Item"

"Handover Date"

"Implementation Plan"

"Key Milestone Date"

### "Milestone Payment"

### means

- a) a delay in the Achievement of a Milestone by its Milestone Date; or
- b) a delay in the design, development, testing or implementation of a Deliverable by the relevant date set out in the Implementation Plan;

means an item or feature in the supply of the Deliverables delivered or to be delivered by the Supplier at or before a Milestone Date listed in the Implementation Plan;

means the date on which the Incumbent Provider hands over services and/or activities back to the Buyer or another Supplier named by the Buyer;

means the set of planning tools (ranging from a traditional Gannt chart through to Agile tools such as Roadmaps, EPIC boards, etc) which may be employed to plan implementation, and includes the Transition Plan;

means a Milestone Date which, if not met, may result in liabilities or Delay Payments;

means a payment identified in the Implementation Plan to be made following the issue of a Satisfaction Certificate in respect of Achievement of the relevant Milestone;

"Mobilisation Date"	means the date on which individual Supplier workers are required to start Buyer related work as identified in an appropriate mobilisation plan;
"Transition Period"	has the meaning given to it in Paragraph 8.2;
"Transition Plan"	means the plan used to transfer activities and/or Services from the Supplier to the Buyer or from the Supplier to another supplier nominated by the Buyer.

### 2. THE IMPLEMENTATION PLAN WITHIN THE CONTEXT OF AGILE

- 2.1 Agile development allows for greater refinement and iteration during development and therefore implementation may involve a diverse set of tools, including but not limited to:
  - Roadmaps (of different shapes and sizes);
  - · EPIC Boards;
  - VMOST Mission Boards; and
  - a wide variety of more granular visual techniques such as Sprint Boards, Kanban Card Walls, etc.
- 2.2 In addition to, or as an alternative to, the traditional Gantt chart type plan, the Buyer may request one or more of the Agile mechanisms described at paragraph 2.1 as part of an Implementation Plan. Annex 3 of this Part A provides a list of the planning tools which may be requested by the Buyer.
- 2.3 The provisions of this Schedule shall apply regardless of any planning or implementation tools requested by the Buyer.

### 3. AGREEING AND FOLLOWING THE IMPLEMENTATION PLAN

- 3.1 A draft of the Implementation Plan for the Call-Off Contract is set out at Annex 1 to this Schedule. The Supplier shall provide a further draft Implementation Plan with the number of days after the Call-Off Contract Start Date specified within Framework Schedule 6A (Health Order Form) under the heading of Further Implementation Plan.
- 3.2 Each Statement of Work may include an Implementation Plan and Milestones specific to that Statement of Work. This shall form a subset of the Call-Off Contract Implementation Plan at a more detailed level of granularity and shall be as set out in the relevant Statement of Work.
- 3.3 This Schedule shall apply, where relevant, to any Implementation Plan regardless of whether at Call-Off Contract level or Statement of Work level.

- 3.4 The draft Implementation Plan:
  - 3.4.1 must contain information at the level of detail necessary to manage the implementation stage effectively and as the Buyer may otherwise require; and
  - 3.4.2 shall take account of all dependencies known to, or which should reasonably be known to, the Supplier.
- 3.5 Following receipt of the draft Implementation Plan from the Supplier, the Parties shall use reasonable endeavours to agree the contents of the Implementation Plan. If the Parties are unable to agree the contents of the Implementation Plan within twenty (20) Working Days of its submission, then such Dispute shall be resolved in accordance with the Dispute Resolution Procedure.
- 3.6 The Supplier shall aim to provide each of the Deliverable Items identified in the Implementation Plan by the date assigned to that Deliverable Item in the Implementation Plan with the aim of ensuring that each Milestone identified in the Implementation Plan is Achieved on or before its Milestone Date.
- 3.7 The Supplier shall monitor its performance against the Implementation Plan and Milestones (if any) and report to the Buyer on such performance.
- 3.8 The Buyer shall identify any Key Milestone Dates which, if missed, will result in liabilities or Delay Payments being incurred. The Key Milestone Dates are as laid out in Annex 2.
- 3.9 The Supplier shall ensure that the critical path leading to any such Key Milestone Dates is clearly identified. In the event that planned Milestone Dates which lie on such a critical path are missed the Supplier shall take appropriate action to rectify and/or recover progress (which may include completion of a Rectification Plan).

### 4. REVIEWING AND CHANGING THE IMPLEMENTATION PLAN

- 4.1 Subject to Paragraph 4.3, the Supplier shall keep the Implementation Plan under review in accordance with the Buyer's instructions and ensure that it is updated on a regular basis.
- 4.2 The Buyer shall have the right to require the Supplier to include any reasonable changes or provisions in each version of the Implementation Plan.
- 4.3 If operating under the Fixed Price model, changes to any Milestones, Milestone Payments and Delay Payments shall only be made in accordance with the Variation Procedure.
- 4.4 Under the Incremental Fixed Price model, it is anticipated that Milestones and Milestone Payments will be refined up to the point of being fixed. Once fixed, changes to Milestones and Milestone Payments shall only be made in accordance with the Variation Procedure.
- 4.5 Under both the Incremental Fixed Price and Capped Time and Materials models, changes to Key Milestone Dates shall only be made in accordance with the Variation Procedure.

4.6 Time in relation to compliance with the Implementation Plan shall be of the essence and failure of the Supplier to meet any Key Milestone Dates shall be a material Default.

### 5. SECURITY REQUIREMENTS BEFORE ANY MOBILISATION DATES

- 5.1 The Supplier shall note that it is incumbent upon them to understand the leadin period for security clearances and ensure that all Supplier Staff have the necessary security clearance in place before any identified Mobilisation Date.
- 5.2 Prior to a team commencing activity (for example at the beginning of a Statement of Work), the Supplier shall ensure that this requirement is reflected in their Implementation Plans.
- 5.3 The Supplier shall ensure that all Supplier Staff and Subcontractors do not access the Buyer's IT systems, or any IT systems linked to the Buyer, unless they have satisfied the Buyer's security requirements.
- 5.4 The Supplier shall be responsible for providing all necessary information to the Buyer to facilitate security clearances for Supplier Staff and Subcontractors in accordance with the Buyer's requirements.
- 5.5 The Supplier shall provide the names of all Supplier Staff and Subcontractors and inform the Buyer of any alterations and additions as they take place throughout the Call-Off Contract.
- 5.6 The Supplier shall ensure that all Supplier Staff and Subcontractors requiring access to the Buyer Premises have the appropriate security clearance. It is the Supplier's responsibility to establish whether or not the level of clearance will be sufficient for access. Unless prior approval has been received from the Buyer, the Supplier shall be responsible for meeting the costs associated with the provision of security cleared escort services.
- 5.7 If a property requires Supplier Staff or Subcontractors to be accompanied by the Buyer's Authorised Representative, the Buyer must be given reasonable notice of such a requirement, except in the case of emergency access.

### WHAT TO DO IF THERE IS A DELAY

- 6.1 If the Supplier becomes aware that there is, or there is reasonably likely to be, a Delay under this Contract it shall:
  - 6.1.1 notify the Buyer as soon as practically possible and no later than within two (2) Working Days from becoming aware of the Delay or anticipated Delay;
  - 6.1.2 include in its notification an explanation of the actual or anticipated impact of the Delay;
  - 6.1.3 comply with the Buyer's instructions in order to address the impact of the Delay or anticipated Delay; and
  - 6.1.4 use all reasonable endeavours to eliminate or mitigate the consequences of any Delay or anticipated Delay.

### 7. COMPENSATION FOR A DELAY

- 7.1 If Delay Payments have been included in the Implementation Plan and a Milestone has not been Achieved by the relevant Key Milestone Date, the Supplier shall pay to the Buyer such Delay Payments (calculated as set out by the Buyer in the Implementation Plan) and the following provisions shall apply:
  - 7.1.1 the Supplier acknowledges and agrees that any Delay Payment is a price adjustment and not an estimate of the Loss that may be suffered by the Buyer as a result of the Supplier's failure to Achieve the corresponding Milestone;
  - 7.1.2 if included within the Implementation Plan, Delay Payments shall be the Buyer's exclusive financial remedy for the Supplier's failure to Achieve Milestones by the relevant Key Milestone Date(s) except where:
    - the Buyer is otherwise entitled to or does terminate this Contract pursuant to Clause 10.4 (When CCS or the Buyer can end this contract); or
    - (b) the delay exceeds the number of days (the "Delay Period Limit") specified in the Implementation Plan commencing on the relevant Key Milestone Date;
  - 7.1.3 the Delay Payments will accrue on a daily basis from the relevant Key Milestone Date until the date when the Milestones leading to that Key Milestone Date are Achieved;
  - 7.1.4 no payment or other act or omission of the Buyer shall in any way affect the rights of the Buyer to recover the Delay Payments or be deemed to be a waiver of the right of the Buyer to recover any such damages; and
  - 7.1.5 Delay Payments shall not be subject to or count towards any limitation on liability set out in Clause 11 (How much you can be held responsible for).

### 8. TRANSITION PLAN

Applies if a Transition Plan is called for within Framework Schedule 6A (Order Form)

- 8.1 The Transition Plan forms part of the overall Implementation Plan
- 8.2 The Transition Period will be a [six (6)] Month period.
- 8.3 During the Transition Period, the Incumbent Provider shall retain full responsibility for all existing activities and Services until the Handover Date or as otherwise formally agreed with the Buyer. The Supplier's full service obligations shall formally be assumed on the Handover Date as set out in the Order Form.

- 8.4 In accordance with the Transition Plan, the Supplier shall:
  - 8.4.1 work cooperatively and in partnership with the Buyer, the Incumbent Provider, and other suppliers, where applicable, to understand the scope of Services to ensure a mutually beneficial handover of the Services;
  - 8.4.2 work with the Incumbent Provider and Buyer to assess the scope of the Services and prepare a plan which demonstrates how they will mobilise the Services;
  - 8.4.3 liaise with the Incumbent Provider to enable the full completion of the Transition Period activities; and
  - 8.4.4 produce a Transition Plan, to be agreed by the Buyer, for carrying out the requirements within the Transition Period including Key Milestones and dependencies.
- 8.5 The Transition Plan will include detail stating:
  - 8.5.1 how the Supplier will work with the Incumbent Provider and the Buyer to capture, transfer and load up information such as software, documentation, pertinent knowledge, data and other information; and
  - 8.5.2 a communications plan, as appropriate and as requested by the Buyer, to be produced and implemented by the Supplier, but to be agreed with the Buyer, including the frequency, responsibility for and nature of communication with the Buyer and end users of the Services.
- 8.6 In addition, the Supplier shall:
  - 8.6.1 appoint a Supplier Authorised Representative who shall be responsible for the management of the Transition Plan, to ensure that the Transition Period is planned and resourced adequately, and who will act as a point of contact for the Buyer;
  - 8.6.2 mobilise all the Services specified in the Specification within the Call-Off Contract;
  - 8.6.3 if appropriate and specifically requested by the Buyer, produce a Transition Plan report for each Buyer Premises to encompass activities that will fulfil all the Buyer's obligations to landlords and other tenants:
    - (a) the format of reports and programmes shall be in accordance with the Buyer's requirements and particular attention shall be paid to establishing the operating requirements of the occupiers when preparing these activities which are subject to the Buyer's approval; and
    - (b) the Parties shall use reasonable endeavours to agree the contents of the report but if the Parties are unable to agree the contents within twenty (20) Working Days of its submission by the Supplier to the Buyer, then such

> Dispute shall be resolved in accordance with the Dispute Resolution Procedure.

- 8.6.4 manage and report progress against the Transition Plan;
- 8.6.5 construct and maintain a Transition Period risk and issue register in conjunction with the Buyer detailing how risks and issues will be effectively communicated to the Buyer in order to mitigate them;
- 8.6.6 attend progress meetings (frequency of such meetings shall be as set out in the Order Form) in accordance with the Buyer's requirements during the Transition Period. Transition meetings shall be chaired by the Buyer and all meeting minutes shall be kept and published by the Supplier; and
- 8.6.7 ensure that all risks associated with the Transition Period are minimised to ensure a seamless change of control between Incumbent Provider and the Supplier.

# Annex 1: Call-Off Contract Implementation Plan

Refer to individual Statements of Work for Implementation Plans specific to those Statements of Work.

Note that Key Milestone Dates (which have a commercial impact if not Achieved) are set out in Annex 2. Key Milestone Dates are Achieved on completion of several non-critical Milestones.

The Call-Off Contract Implementation Plan is set out below and the Milestones to be Achieved are identified below:

Milestone	Deliverable Items	Milestone Date	Buyer Responsibilities	Reference to Acceptance Criteria	Milestone Payment (if applicable)	Applicable Delay Payment(s)	
2		[]	[]	2	[]	0	z

I ne milestones will be Achieved in accordance with this Schedule.

Framework Ref: RM6221 Project Version: v2.0 Model Version: v3.2

113 (22/12/2021)

80

## Annex 2: Key Milestone Dates

All Milestones listed in Annex 1 with a Milestone Date which precedes the Key Milestone Date listed below are required to be completed and accepted in order for a Key Milestone Date to be considered to have been met

		Milestone Date	Payments	Kequired
[] []	2	[]	[]	z

Framework Ref: RM6221 Project Version: v2.0 Model Version: v3.2

114 (22/12/2021)

σ

### Annex 3: Planning Tools

The following table provides a list of planning tools which may be required by the Buyer.

### Notes:

- This list is non-exclusive, and the Buyer may elect to include other information.
  This list is also illustrative and non-exhaustive. It is anticipated the Buyer will review and edit these tables as part of preparing a Call-Off Contract (and reserves the right to amend during the Call-Off Contract Period)

Ref.	Type of Information	Required?	Refresh Frequency
Tradition	Traditional Planning Artifacts	Å	25 25 25 25 25 25 25 25 25 25 25 25 25 2
A6.01	Call-Off Contract Plan on a Page. A high level plan covering the duration and scope of the Call-Off Contract	z	Commercial Planning / Review Event
A6.02	SOW Plan on a Page. A high level plan covering the duration and scope of an individual SOW	z	Operational Planning Event
A6.03	Full duration Project Plan including resources, dependencies, etc (e.g. as created by traditional project planning software)	z	Operational Planning Event
A6.04	Rolling 3-Month Detailed Look Ahead Plan (as created by traditional project planning software)	z	Operational Planning Event
Agile Pla	Agile Planning Artifacts		_
A6.10	Product Road-Map	z	On request from Buyer
A6.11	Delivery Plans	z	On request from Buyer

Framework Ref: RM6221 Project Version: v2.0 Model Version: v3.2 C52340 NHS Pathways 03 Call Off Schedules 1.0F 22Dec21

### Call-Off Schedule 13A: (Health Implementation Plan and Testing) Call-Off Ref: C52340 PD - NHS Pathways Crown Copyright 2018

Ref.	Type of Information	Required?	Refresh Frequency
A6.12	Timebox Plans	z	On request from Buyer
A6.13	EPIC Board	z	On request from Buyer
A6.14	Elaboration Board	Z	On request from Buyer
A6.15	Sprint Board	N	On request from Buyer
A6.16	Agile Kanban Board	z	On request from Buyer

Framework Ref: RM6221 Project Version: v2.0 Model Version: v3.2

116 (22/12/2021)

1

### PART B - TESTING

### 1. DEFINITIONS

1.1 In this Schedule, the following words shall have the following meanings and they shall supplement Joint Schedule 1 (Definitions):

> "Component" means any constituent parts of the Deliverables: means a test issue of Severity Level 1 "Material Test Issue" or Severity Level 2 as set out in the relevant Test Plan: "Release" shall mean a set of Deliverables, agreed by the Buyer as being something which can be released to the intended users of the set of Deliverables: "Satisfaction means a certificate materially in the Certificate" form of the document contained in Annex 2: issued by the Buyer when a Deliverable and/or Milestone has satisfied its relevant Test Success Criteria: Severity Level means the level of severity of a Test Issue, the criteria for which are described in Annex 1:: 'Test' means a procedure intended to establish the quality, performance, or reliability of a Component; "Test Issue" means a test issue of Severity Level 3, Severity Level 4 or Severity Level 5 as set out in the relevant Test Plan; "Test Issue means a log for the recording of Test Management Log" Issues as described further in Paragraph 9.1 of this Schedule; "Test Plan" means a specific set of tests to be incorporated as part of a release plan; "Test Reports" means the reports to be produced by the Supplier setting out the results of Tests:

C52340 NHS Pathways 03 Call Off Schedules 1.0F 22Dec21

of Rainbow Teams or Self Directed Teams accountability for testing shall be clearly defined within the Test Strategy (since it may involve both Buyer and Supplier). Accountability may vary from Statement of Work to

applicable

procedures and Test Success Criteria

testing

Statement of Work; and

the

set out in this Schedule.

means

Call-Off Schedule 13A: (Health Implementation Plan and Testing) Call-Off Ref: C52340 PD - NHS Pathways Crown Copyright 2018

"Test Specification"	means the specification that sets out how Tests will demonstrate that the Test Success Criteria have been satisfied, as described in more detail in Paragraph 7.2 of this Schedule;
"Test Strategy"	means a strategy for the conduct of Testing as described further in Paragraph 4.2 of this Schedule;
"Test Success Criteria"	in relation to a Test, the test success criteria for that Test as referred to in Paragraph 6 of this Schedule;
"Test Witness"	means any person appointed by the Buyer pursuant to Paragraph 10 of this Schedule;
"Tester"	means the organisation responsible for testing. Within the context of Sole Responsibility accountability this shall mean the Supplier. Within the context

"Testing Procedures"

### 2. AGILE TESTING

- 2.1 Testing under this Schedule shall apply to software as well as other Deliverables (for example migration of data sets).
- 2.2 When requested by the Buyer, the Supplier shall, with appropriate input from the Buyer, undertake Release planning and shall produce a Test Plan.
- 2.3 With each Agile development iteration development teams are required to incorporate the underlying ethos of Agile testing and the Supplier shall:
  - (i) write the test script;
  - (ii) write the code / perform the digital activity; and
  - (iii) test the code / activity against the test script.
- 2.4 Wherever possible Testers are required to automate testing (e.g. relevant functional test cases) and automate and execute any regression tests. This shall form part of acceptance testing.
- 2.5 Prior to release additional tests such as, but not limited to:
  - (i) load tests;
  - (ii) complete regression tests;
  - (iii) penetration tests;
  - (iv) user acceptance tests;
  - (v) integration tests;
  - (vi) deployment tests; and
  - (vii) Release readiness tests

as agreed within the Test Plan, shall be executed by the Tester.

- 2.6 The Supplier shall develop reusable test scripts in a modular manner which can be incorporated within a larger library of routinely run test scripts.
- 2.7 The Test Strategy will define how testing within the context of iterative agile development interacts with the broader testing at Deliverable, Statement of Work and Call-Off Contract levels.

### 3. HOW TESTING SHOULD WORK

- 3.1 All Tests conducted by the Tester shall be conducted in accordance with the Test Strategy, Test Specification and, as appropriate within an agile context, the Test Plan.
- 3.2 The Tester shall not submit any Deliverable for Testing:
  - 3.2.1 unless the Tester is reasonably confident that it will satisfy the relevant Test Success Criteria;
  - 3.2.2 until the Buyer has issued a Satisfaction Certificate in respect of any prior, dependant Deliverable(s); and
  - 3.2.3 until the Parties have agreed the Test Plan and the Test Specification relating to the relevant Deliverable(s).
- 3.3 Where agreed within the Test Plan, the Tester shall use reasonable endeavours to submit each Deliverable for Testing or re-Testing by or before the date set out in the Implementation Plan for the commencement of Testing in respect of the relevant Deliverable.
- 3.4 Agile iterative tests, as agreed within the Test Strategy, shall align with the iterative development cycle and shall contribute to, but not necessarily be reflected in, the Deliverable level testing regime described within this Schedule.
- 3.5 Prior to the issue of a Satisfaction Certificate, the Buyer shall be entitled to review the relevant Test Reports and the Test Issue Management Log.

### 4. PLANNING FOR TESTING

- 4.1 As part of the first relevant Statement of Work under the Call-Off Contract, the Supplier shall develop a relevant Test Strategy as soon as practicable after the SOW Start Date but in any case no later than twenty (20) Working Days after the SOW Start Date.
- 4.2 The initial Test Strategy shall include:
  - 4.2.1 an overview of how Testing will be conducted in relation to the Implementation Plan relevant to the Statement of Work;
  - 4.2.2 the process to be used to capture and record Test results and the categorisation of Test Issues;
  - 4.2.3 the procedure to be followed should a Deliverable fail a Test, fail to satisfy the Test Success Criteria or where the Testing of a Deliverable produces unexpected results, including a procedure for the resolution of Test Issues;
  - 4.2.4 the procedure to be followed to sign off each Test;
  - 4.2.5 the process for the production and maintenance of Test Reports and a sample plan for the resolution of Test Issues;

- 4.2.6 the names and contact details of the Buyer and the Supplier's Test representatives;
- 4.2.7 a high level identification of the resources required for Testing including Buyer and/or third party involvement in the conduct of the Tests;
- 4.2.8 the technical environments required to support the Tests; and
- 4.2.9 the procedure for managing the configuration of the Test environments.
- 4.3 The Test Strategy shall be approved by the Buyer.
- 4.4 As new Statements of Work are initiated, the Test Strategy is required to be refined to reflect the specific needs of each Statement of Work and shall reflect the Test Strategy required under the Call-Off Contract as a whole.

### 5. PREPARING FOR TESTING

- 5.1 As defined within the Test Strategy, the Tester shall develop Test Plans and submit these for Approval as soon as practicable but in any case no later than twenty (20) Working Days prior to the start date for the relevant Testing as specified in the relevant Implementation Plan.
- 5.2 Each Test Plan shall include as a minimum:
  - 5.2.1 the relevant Test definition and the purpose of the Test, the Milestone to which it relates, the requirements being Tested and, for each Test, the specific Test Success Criteria to be satisfied; and
  - 5.2.2 a detailed procedure for the Tests to be carried out.
- 5.3 The Buyer shall not unreasonably withhold or delay its approval of the Test Plan provided that the Tester shall implement any reasonable requirements of the Buyer in the Test Plan.

### 6. PASSING TESTING

6.1 The Test Success Criteria for all Tests shall be agreed between the Parties as part of the relevant Test Plan pursuant to Paragraph 5.

### 7. HOW DELIVERABLES WILL BE TESTED

7.1 Following approval of a Test Plan, the Tester shall develop the Test Specification for the relevant Deliverables as soon as reasonably practicable and in any event at least 10 Working Days prior to the start of the relevant Testing (as specified in the Implementation Plan).

- 7.2 Each Test Specification shall include as a minimum:
  - 7.2.1 the specification of the Test data, including its source, scope, volume and management, a request (if applicable) for relevant Test data to be provided by the Buyer and the extent to which it is equivalent to live operational data;
  - 7.2.2 a plan to make the resources available for Testing;
  - 7.2.3 Test scripts;
  - 7.2.4 Test pre-requisites and the mechanism for measuring them; and
  - 7.2.5 expected Test results, including:
    - (a) a mechanism to be used to capture and record Test results; and
    - (b) a method to process the Test results to establish their content.

### 8. PERFORMING THE TESTS

- 8.1 Before submitting any Deliverables for Testing the Tester shall subject the relevant Deliverables to its own internal quality control measures.
- 8.2 The Tester shall manage the progress of Testing in accordance with the relevant Test Plan and shall carry out the Tests in accordance with the relevant Test Specification. Tests may be witnessed by the Test Witnesses in accordance with Paragraph 10.3.
- 8.3 The Tester shall notify the Buyer at least 10 Working Days in advance of the date, time and location of the relevant Tests and the Buyer shall ensure that the Test Witnesses attend the Tests.
- 8.4 The Buyer may raise and close Test Issues during the Test witnessing process.
- 8.5 The Tester shall provide to the Buyer in relation to each Test:
  - 8.5.1 a draft Test Report not less than 2 Working Days prior to the date on which the Test is planned to end; and
  - 8.5.2 the final Test Report within 5 Working Days of completion of Testing.

- 8.6 Each Test Report shall provide a full report on the Testing conducted in respect of the relevant Deliverables, including:
  - 8.6.1 an overview of the Testing conducted;
  - 8.6.2 identification of the relevant Test Success Criteria that have/have not been satisfied together with the Tester's explanation of why any criteria have not been met;
  - 8.6.3 the Tests that were not completed together with the Tester's explanation of why those Tests were not completed;
  - 8.6.4 the Test Success Criteria that were satisfied, not satisfied or which were not tested, and any other relevant categories, in each case grouped by Severity Level in accordance with Paragraph 9.1; and
  - 8.6.5 the specification for any hardware and software used throughout Testing and any changes that were applied to that hardware and/or software during Testing.
- 8.7 When a Milestone has been completed the Tester shall submit any Deliverables relating to that Milestone for Testing.
- 8.8 Each party shall bear its own costs in respect of the Testing. However, if a Milestone is not Achieved the Buyer shall be entitled to recover from the Supplier any reasonable additional costs it may incur as a direct result of further review or re-Testing of a Milestone.
- 8.9 If the Tester successfully completes the requisite Tests, the Buyer shall issue a Satisfaction Certificate as soon as reasonably practical following such successful completion.
- 8.10 Notwithstanding the issuing of any Satisfaction Certificate, the Supplier shall remain responsible for ensuring that the Testing of Deliverables are implemented in accordance with this Contract as defined within the Test Strategy.

### 9. DISCOVERING PROBLEMS

- 9.1 Where a Test Report identifies a Test Issue, the Parties shall agree the classification of the Test Issue using the criteria specified in Annex 1: and the Test Issue Management Log maintained by the Tester shall log Test Issues reflecting the Severity Level allocated to each Test Issue.
- 9.2 The Tester shall be responsible for maintaining the Test Issue Management Log and for ensuring that its contents accurately represent the current status of each Test Issue at all relevant times. The Tester shall make the Test Issue Management Log available to the Buyer upon request.
- 9.3 The Buyer shall confirm the classification of any Test Issue unresolved at the end of a Test in consultation with the Tester. If the Parties are unable to agree the classification of any unresolved Test Issue, the Dispute shall be dealt with in accordance with the Dispute Resolution Procedure using an expedited dispute timetable.

### 10. TEST WITNESSING

- 10.1 The Buyer may, in its sole discretion, require the attendance at any Test of one or more Test Witnesses selected by the Buyer, each of whom shall have appropriate skills to fulfil the role of a Test Witness.
- 10.2 The Tester shall give the Test Witnesses access to any documentation and Testing environments reasonably necessary and requested by the Test Witnesses to perform their role as a Test Witness in respect of the relevant Tests.
- 10.3 The Test Witnesses:
  - 10.3.1 shall actively review the Test documentation;
  - 10.3.2 will attend and engage in the performance of the Tests on behalf of the Buyer so as to enable the Buyer to gain an informed view of whether a Test Issue may be closed or whether the relevant element of the Test should be re-Tested;
  - 10.3.3 shall not be involved in the execution of any Test;
  - 10.3.4 shall be required to verify that the Tester conducted the Tests in accordance with the Test Success Criteria and the relevant Test Plan and Test Specification;
  - 10.3.5 may produce and deliver their own, independent reports on Testing, which may be used by the Buyer to assess whether the Tests have been Achieved;
  - 10.3.6 may raise Test Issues on the Test Issue Management Log in respect of any Testing; and
- 10.4 may require the Tester to demonstrate the modifications made to any defective Deliverable before a Test Issue is closed.

### 11. AUDITING THE QUALITY OF THE TEST

- 11.1 The Buyer or an agent or contractor appointed by the Buyer may perform ongoing quality audits in respect of any part of the Testing (each a "Testing Quality Audit").
- 11.2 The Tester shall allow sufficient time in the Test Plan to ensure that adequate responses to a Testing Quality Audit can be provided.
- 11.3 The Buyer will give the Tester at least 5 Working Days' written notice of the Buyer's intention to undertake a Testing Quality Audit.
- 11.4 The Tester shall provide all reasonable necessary assistance and access to all relevant documentation required by the Buyer to enable it to carry out the Testing Quality Audit.
- 11.5 If the Testing Quality Audit gives the Buyer concern in respect of the Testing Procedures or any Test, the Buyer shall prepare a written report for the Tester detailing its concerns and the Tester shall, within a reasonable timeframe, respond in writing to the Buyer's report.

11.6 In the event of an inadequate response to the written report from the Tester, the Buyer (acting reasonably) may withhold a Satisfaction Certificate until the issues in the report have been addressed to the reasonable satisfaction of the Buyer.

### 12. OUTCOME OF THE TESTING

- 12.1 The Buyer will issue a Satisfaction Certificate when the Deliverables satisfy the Test Success Criteria in respect of that Test without any Test Issues.
- 12.2 If the Deliverables (or any relevant part) do not satisfy the Test Success Criteria then the Buyer shall notify the Tester and:
  - 12.2.1 the Buyer may issue a Satisfaction Certificate conditional upon the remediation of the Test Issues;
  - 12.2.2 the Buyer may extend the Test Plan by such reasonable period or periods as the Parties may reasonably agree and require the Tester to rectify the cause of the Test Issue and re-submit the Deliverables (or the relevant part) to Testing; or
  - 12.2.3 where the Supplier has Sole Responsibility and where the failure to satisfy the Test Success Criteria results, or is likely to result, in the failure (in whole or in part) by the Supplier to meet a Milestone, then without prejudice to the Buyer's other rights and remedies, such failure shall constitute a material Default.
- 12.3 Where the Supplier has Sole Responsibility, the Buyer shall be entitled, without prejudice to any other rights and remedies that it has under this Contract, to recover from the Supplier any reasonable additional costs it may incur as a direct result of further review or re-Testing which is required for the Test Success Criteria for that Deliverable to be satisfied.
- 12.4 The Buyer shall issue a Satisfaction Certificate in respect of a given Milestone as soon as is reasonably practicable following:
  - 12.4.1 the issuing by the Buyer of Satisfaction Certificates and/or conditional Satisfaction Certificates in respect of all Deliverables related to that Milestone which are due to be Tested; and
  - 12.4.2 performance by the Supplier to the reasonable satisfaction of the Buyer of any other tasks identified in the Implementation Plan as associated with that Milestone.
- 12.5 The grant of a Satisfaction Certificate shall entitle the Supplier to the receipt of a payment in respect of that Milestone in accordance with the provisions of any Implementation Plan and Clause 4 (Pricing and payments).
- 12.6 If a Milestone is not Achieved, the Buyer shall promptly issue a report to the Supplier setting out the applicable Test Issues and any other reasons for the relevant Milestone not being Achieved.
- 12.7 If there are Test Issues, but these do not exceed the Test Issues Threshold, then provided there are no Material Test Issues, the Buyer shall issue a Satisfaction Certificate.

- 12.8 If there is one or more Material Test Issue(s), the Buyer shall refuse to issue a Satisfaction Certificate and, without prejudice to the Buyer's other rights and remedies, such failure shall constitute a material Default should the Supplier hold Sole Responsibility
- 12.9 If there are Test Issues which exceed the Test Issues Threshold but there are no Material Test Issues, the Buyer may at its discretion (without waiving any rights in relation to the other options) choose to issue a Satisfaction Certificate conditional on the remediation of the Test Issues in accordance with an agreed Rectification Plan provided that:
  - 12.9.1 any Rectification Plan shall be agreed before the issue of a conditional Satisfaction Certificate unless the Buyer agrees otherwise (in which case the Tester shall submit a Rectification Plan for approval by the Buyer within 10 Working Days of receipt of the Buyer's report pursuant to Paragraph 11.5); and
  - 12.9.2 where the Buyer issues a conditional Satisfaction Certificate, it may (but shall not be obliged to) revise the failed Milestone Date and any subsequent Milestone Date.

### 13. RISK

- 13.1 The issue of a Satisfaction Certificate and/or a conditional Satisfaction Certificate shall not:
  - 13.1.1 operate to transfer any risk that the relevant Deliverable or Milestone is complete or will meet and/or satisfy the Buyer's requirements for that Deliverable or Milestone; or
  - 13.1.2 affect the Buyer's right subsequently to reject all or any element of the Deliverables and/or any Milestone to which a Satisfaction Certificate relates.
# Annex 1:

# Test Issues – Severity Levels

## 1. SEVERITY 1 ERROR

1.1 This is an error that causes non-recoverable conditions, e.g. it is not possible to continue using a Component.

## 2. SEVERITY 2 ERROR

- 2.1 This is an error for which, as reasonably determined by the Buyer, there is no practicable workaround available, and which:
  - 2.1.1 causes a Component to become unusable;
  - 2.1.2 causes a lack of functionality, or unexpected functionality, that has an impact on the current Test; or
  - 2.1.3 has an adverse impact on any other Component(s) or any other area of the Deliverables;

## 3. SEVERITY 3 ERROR

- 3.1 This is an error which:
  - 3.1.1 causes a Component to become unusable;
  - 3.1.2 causes a lack of functionality, or unexpected functionality, but which does not impact on the current Test; or
  - 3.1.3 has an impact on any other Component(s) or any other area of the Deliverables;

but for which, as reasonably determined by the Buyer, there is a practicable workaround available;

## 4. SEVERITY 4 ERROR

4.1 This is an error which causes incorrect functionality of a Component or process, but for which there is a simple, Component based, workaround, and which has no impact on the current Test, or other areas of the Deliverables.

## 5. SEVERITY 5 ERROR

5.1 This is an error that causes a minor problem, for which no workaround is required, and which has no impact on the current Test, or other areas of the Deliverables. Call-Off Schedule 13A: (Health Implementation Plan and Testing) Call-Off Ref: C52340 PD - NHS Pathways Crown Copyright 2018

# Annex 2:

# Satisfaction Certificate

To: [insert name of Supplier]

From: [insert name of Buyer]

[insert Date dd/mm/yyyy]

Dear Sirs,

### Satisfaction Certificate

Deliverable/Milestone(s): [Insert relevant description of the agreed Deliverables/Milestones].

We refer to the agreement ("Call-Off Contract") [insert Call-Off Contract reference number] relating to the provision of the [insert description of the Deliverables] between the [insert Buyer name] ("Buyer") and [insert Supplier name] ("Supplier") dated [insert Call-Off Start Date dd/mm/yyyy].

The definitions for any capitalised terms in this certificate are as set out in the Call-Off Contract.

[We confirm that all the Deliverables relating to [insert relevant description of Deliverables/agreed Milestones and/or reference number(s) from the Implementation Plan] have been tested successfully in accordance with the Test Plan [or that a conditional Satisfaction Certificate has been issued in respect of those Deliverables that have not satisfied the relevant Test Success Criteria].

# [OR]

[This Satisfaction Certificate is granted on the condition that any Test Issues are remedied in accordance with the Rectification Plan attached to this certificate.]

[You may now issue an invoice in respect of the Milestone Payment associated with this Milestone in accordance with Clause 4 (Pricing and payments)].

Yours faithfully

[insert Name]

[insert Position]

acting on behalf of [insert name of Buyer]

# Call-Off Schedule 15A (Health Supplier and Contract Management)

### 1. DEFINITIONS

1.1 In this Schedule, the following words shall have the following meanings and they shall supplement Joint Schedule 1 (Definitions):

"Agile"	a generic term to cover agile ways of working within the digital environment;
"Backlog"	has the meaning given to it in paragraph 2.3.1;
"Balanced Scorecard"	has the meaning given to it in paragraph 2.3.8;
"Buyer SM Event"	has the meaning given to it in paragraph 2.3.7;
"Call-Off Contract Management"	has the meaning given to it in paragraph 2.1.2;
"Call-Off Rate Card"	means the table of rates for different roles as captured in Call-Off Schedule 5A (Pricing Details);
"Collaborative Buyer/Supplier Event"	has the meaning given to it in paragraph 2,3.9;
"Call-Off Contract Manager"	the Call-Off Contract Manager appointed for the Supplier and for the Buyer in accordance with Annex 6 of this Schedule;
"Commercial Planning/Review Events"	has the meaning given to it in paragraph 2.3.4;
"Executive Sponsor"	has the meaning given to it in paragraph 6.26;
"Operational Board"	the board established in accordance with paragraph 2.3.64.1 of this Schedule;
"Operational Contract Manager"	the operational contract manager appointed for the Supplier and for the Buyer in Annex 6 of this Schedule;
"Operational Planning Event"	has the meaning given to it in paragraph 2.3.2;
"Operational Review Events"	has the meaning given to it in paragraph 2.3.3;
"Orange Book"	means the Orange Book: Management of Risks – Principles and Concepts accessible at: https://assets.publishing.service.gov.uk/governme nt/uploads/system/uploads/attachment_data/file/1 91513/The Orange Book.pdf
"Programme"	means a programme of work, as identified by a portfolio number or equivalent;

"Scrum Sprint"	means a time boxed period of time in which a useable product increment is created per the
"SOW Management"	Scrum Guide <sup>™</sup> (an Agile methodology); has the meaning given to it in paragraph 2.1.1; and
"SOW Variation"	has the meaning given to it in paragraph 2.3.5.

#### 2. OVERVIEW OF SUPPLIER AND CONTRACT MANAGEMENT

- 2.1 There are five levels of relationship and contract management covered by the various parts of this framework:
  - 2.1.1 Statement of Work Management ("SOW Management") covering how Statements of Work (SOWs) are managed. SOW Management starts with SOWs being created and ends, typically with handover to the next SOW, as a SOW ends (see paragraph 3.1);
  - 2.1.2 Call-Off Contract Management ("Call-Off Contract Management") covering the specific contract management obligations between Supplier and Buyer relating to a specific Call-Off Contract (see paragraph 3.8);
  - 2.1.3 Buyer/Supplier Management covering individual relationships between the Buyer and a single Supplier. These relationships are anticipated to develop as an early part of Call-Off competition as Suppliers bid for potential work. These relationships are also anticipated to persist beyond individual Call-Offs for the duration of the Framework (and beyond, assuming Call-Offs extend beyond the formal term of the Framework) (see paragraph 3.31);
  - 2.1.4 Buyer Specific Framework Management, involving the Buyer and all Suppliers under the Framework with whom the Buyer has a relationship. This level of relationship is anticipated to evolve as Call-Offs are awarded. This level of relationship is anticipated to persist beyond individual Call-Offs (see paragraph 3.40); and
  - 2.1.5 Framework Buyer/Supplier Management covering "user group" topics such as:
    - Innovation;
    - New developments in technologies;
    - Changes in digital services (e.g. the incorporation of SFIA 8 and changes to Digital, Data and Technology roles);
    - · Opportunities to improve the operation of the overall framework; and
    - Proposed updates to the framework;
- 2.2 For the avoidance of doubt, by signing any individual Call-Off Contract, suppliers are obliged to meet the terms of this Schedule, as required by the Buyer, for the durations indicated at each level of relationship.
- 2.3 There are a number of key mechanisms for managing this Call-Off Contract set out below. This list is not an exclusive list but is designed to focus on the key principles underpinning each level of management. Also, whilst logically separated below, the Buyer may elect to combine such mechanisms:
  - 2.3.1 Product Backlog (Scrum <sup>™</sup>) or Prioritised Requirements List (AgilePM<sup>®</sup>) (the "Backlog") which, within the context of this Contract, should be

> considered to be the cornerstone of ongoing operational management. As such it should be considered to be a live contract artefact (see paragraph 4.1);

- 2.3.2 "Operational Planning Events" or equivalent, managing the granular level refinement from Commercial Planning/Review Event level (typically at requirement/Increment level down to Scrum Sprint / Timebox (AgilePM) sized task level (to be held at the frequency given at paragraph 3.6);
- 2.3.3 "Operational Review Events" or equivalent, which provide the basis for among other things, reviewing achievements out of individual Sprints/Timeboxes, learning lessons from the previous activity and understanding the actual effort used. A summary of this will be fed into the Commercial Planning/Review Events (Scrum Sprints or reporting performance and progress of tasks into the Commercial Planning/Review Events is described below at paragraph 3.7);
- 2.3.4 "Commercial Planning/Review Events" (usually combined but may be separate events for planning and review):
  - These events will refine and prioritise the main Backlog (at Deliverable Increment/user story level and above) for feeding into the more detailed SOW task level backlog managed under the Operational Planning Events described above. If the Incremental Fixed Price charging mechanism is in use under a SOW, these events will agree the "price" for a Deliverable Increment (backed up by an estimate of effort linked to the Call-Off Rate Card).
  - From a review perspective, the Commercial Planning/Review Event will assess completeness of tasks and map this into acceptance of Deliverables or Deliverable Increments as well as collate actual effort spent. Acceptance certificates may be used to acknowledge completeness. The review will typically lead to invoicing and payment.
  - Whilst the frequency of Operational Planning Events is likely to reflect the operational practices specific to the SOW, Commercial Planning/Review Events will normally be held on a monthly basis.
  - These events are seen at the main focus points for Call-Off Contract Management activities (see paragraph 3.8).
- 2.3.5 Variation and Change Management covers minor changes to Statements of Work ("SOW Variations") and more formal changes to the contract (see paragraph 4.2 below);
- 2.3.6 Events such as Operational Board meetings, where the programme or project teams may escalate topics for resolution or where material decisions may need to be made. (see paragraph 6.21 below);
- 2.3.7 "Buyer SM Events" designed to aggregate Call-Off Contract activity into an overarching view of Supplier performance and provide a means to take a holistic view of the performance and relationship with the Supplier as well as address topics escalated from individual Call-Off Contracts. If the Buyer only has one or two Call-Off Contracts with an individual supplier these events may be combined with Commercial Planning/Review Events, however the intent is to focus at the Supplier level rather than the Call-Off Contract level (see paragraph 3.39 below);
- 2.3.8 The "Balanced Scorecard", providing the mechanism to visually summarise the status of either the Call-Off and/or Supplier performance. Whilst a large individual Call-Off may warrant a dedicated Balanced

> Scorecard, within the context of this framework it is anticipated the Balanced Scorecard will operate at the overall Buyer/Supplier level (and this is the assumption underpinning the illustrations making up the rest of this Schedule) (see 4.3 below);

- 2.3.9 "Collaborative Buyer/Supplier Events" intended to provide a mechanism to share learning, develop collective improvements, share future plans, and discuss topics which may be forwarded for consideration at the Framework level. (see 3.40 below); and
- 2.3.10 in order to manage the time dimension relevant to the Call-Off Contract the Buyer will require some form of delivery planning. This may be in the form of Gantt charts or project plans, or it may be in the form of agile tools such as roadmaps, Epic boards, Elaboration and/or Sprint Boards, Kanban boards, etc. (see paragraph Error! Reference source not found. below).
- 2.4 Throughout the lifetime of both individual Call-Off Contracts and the framework relationship between the Buyer and Suppliers there are risks. Risk management practices applied at Call-Off Contract level are described further at paragraph 5.
- 2.5 Suppliers should be aware that the Buyer will also undertake ongoing Supplier risk management activities (such as ongoing financial credit rating checks) as part of its Call-Off Contract management processes.

#### 3. SUPPLIER AND CONTRACT MANAGEMENT LEVELS

#### Statement of Work Management

- 3.1 Other than the first SOWs under a Call-Off Contract, which will be defined by the Buyer, SOWs will generally be developed involving joint exploratory discussions between the Buyer and the Supplier. Generally speaking, inputs to the SOW are likely to include:
  - 3.1.1 some form of road map (the context);
  - 3.1.2 initial views on initial resource profile and technology stack and so on, used to inform the sizing of the SOW;
  - 3.1.3 which Accountability Model (Sole Responsibility, Self-Directed Teams or Rainbow), each as defined in Framework Schedule 1 (Specification); and which pricing mechanism (Fixed Price, Capped Time and Material, or Incremental Fixed Priced); and
  - 3.1.4 an initial Backlog (detailing requirements at an appropriate level of detail), provisional resource profile and technologies.
- 3.2 The operation of the Backlog is described under paragraph 4, however, at operational level it is intended to be the basis for agreeing, accepting and capturing tasks as well as related effort. There shall be a continuous alignment between the main Backlog held at Call-Off Contract level and any SOW Backlog (the Call-Off Contract level Backlog focuses on Deliverable Increments whilst the SOW Backlog level focuses on the more detailed tasks). At SOW level the Backlog should largely be operational and should be being refined to a level that can be allocated to the next sprint, timebox or equivalent (aiming to task size).

- 3.3 In addition to the Backlog, there will be information required by the Buyer for the purposes of recording resources (e.g. for controlling access to infrastructure), measuring performance (e.g. burn charts, etc), evidencing delivery (e.g. acceptance certificates) and so on. Annex 1 provides a provisional, non-exclusive, list of information types which may be used by the Buyer to initially specify typical requirements.
- 3.4 Whilst individual SOWs may operate under the Fixed Price or Incremental Fixed Price pricing mechanisms, all Call-Off Contracts shall be underpinned by an agreed set of Supplier Call-Off Rate Cards. Under all circumstances, regardless of pricing model, Suppliers shall maintain an accurate record of actual resource utilisation and to make this available to the Buyer on request.
- 3.5 When capturing effort, the Supplier is required to link such effort to the SOW and to the respective entry on the applicable Call-Off Rate Card as well as include period start and end date and utilised effort (in hours, days or fractions thereof as determined by the Buyer).
- 3.6 Operational Planning Events will be held at a frequency determined by the Buyer. The purpose will be to agree the next iteration of work (e.g. Scrum Sprint) and to refine the Backlog. The principle of ongoing refinement is a key Agile concept.
- 3.7 Operational Review Events will be held at a frequency determined by the Buyer. Typically, these will be at regular intervals (e.g. every month or every Scrum Sprint). Two contractual related purposes of Operational Review Events are to:
  - 3.7.1 identify when tasks are completed and provide evidence to the Commercial Planning/Review Events that work is "done"; and
  - 3.7.2 capture actual effort taken (versus the forecast) as a means of improving future estimates and providing the raw data for invoicing purposes.

#### Call-Off Contract Management

- 3.8 Within the context of this Call-Off Contract, taking into account the other levels of contract management, the primary purpose of Call-Off Contract Management includes:
  - 3.8.1 establishing and managing the information flows relevant to the Call-Off Contract;
  - 3.8.2 managing the overarching Backlog and ensuring it is continuously refined to reflect the evolving work;
  - 3.8.3 establishing new SOWs, providing oversight of SOWs in progress and ensuring handover between SOWs as appropriate;
  - 3.8.4 acting as the bridge between SOW management (operational) and commercial matters such as raising and managing invoices and payment, and tracking and managing commitment and spend against the overall Call-Off Contract value estimate;
  - 3.8.5 providing oversight of the resources (Supplier Staff, Subcontractors, etc.) required to deliver the Deliverables under the Call-Off Contract;
  - 3.8.6 creating and maintaining Implementation Plans (as set out in Call-Off Schedule 13A (Health Implementation Plan and Testing);

- 3.8.7 co-ordinating with stakeholders and the Operational Board if applicable;
- 3.8.8 managing overall Call-Off Contract level risks, issues, escalations and commercial matters;
- 3.8.9 managing formal Variations;
- 3.8.10 acting as the linkage point into Buyer/Supplier Management;
- 3.8.11 maintaining and updating the Contract specific Call-Off Rate Card(s);
- 3.8.12 contributing to the maintenance of programme/project artifacts such as business cases, procurement packages, roadmaps, etc;
- 3.8.13 ensuring smooth transition and hand-over to the recipient of Deliverables (always the Buyer, but it may also be to another supplier if there is another major phase of work to be undertaken by the other supplier under a separate Call-Off Contract or SOW); and
- 3.8.14 monitoring Supplier performance against Material KPIs.
- 3.9 Typically, the Buyer shall expect oversight of SOWs to form part of the role of a core team who will persist for the duration of the Call-Off Contract. The Buyer shall consider executing a dedicated SOW (typically a rainbow team, on a Capped Time and Materials basis) for defining and managing Call-Off Contract Management activities. This would be in addition to any specific delivery SOWs. This role is not anticipated to be full time, but rather periodic (e.g. no more than a few days per month). If a single SOW is operating at any given time, the role may to be combined with others.
- 3.10 Whereas SOWs will often be sequential, there is no restriction on the number of SOWs that are able to be effective at any point in time. Likewise it is highlighted that Service Provisions under this framework are not lots. This means a Call-Off Contract may involve more than one Service Provision (with individual SOWs possibly commissioned to deliver distinct services).
- 3.11 The above activities are logically defined under the heading of Call-Off Contract Management for the purposes of this Call-Off Contract however the Buyer may choose to capture the above requirements in a manner of their own choosing.

#### Establishing and Managing Information Flows

- 3.12 Annex 2 provides an initial list of information which may be requested by the Buyer in connection with Call-Off Contract Management. The Buyer may add to this list at any point in time by notifying the Supplier in writing.
- 3.13 The information set out in Annex 2 shall be expected to be kept up to date by the Supplier at the refresh frequency set out in that Annex. The Supplier is required to maintain tight version control and, where noted, obtain Buyer approval to updates as the work progresses.

#### Managing the Backlog

3.14 The Backlog is a key artifact for Call-Off Contract Management. The Backlog shall track Deliverable Increments as they are refined during the Call-Off Contract Period, will identify which SOWs cover each Deliverable Increment, will size them (and cost them under either the Fixed Price or Incremental Fixed Price models),will provide the basis for "accepting" them and will track various other information as set out in Annex 4.

Crown Copyright 2018

- 3.15 Whilst the Buyer may choose to have a single Backlog covering each SOW and the Call-Off Contract, at Call-Off Contract level the Backlog is intended to capture the Deliverable Increments whilst delegating the detailed task-level management down to the SOW Backlog.
- 3.16 Payment under the either the Fixed Price or Incremental Fixed Price pricing mechanism will be based on the price agreed in advance. However, the Supplier will be expected to provide a resource estimate which, when combined with Call-Off Rate Card prices, will provide the transparent basis for the fixed price.
- 3.17 The Buyer may elect to use acceptance certificates for all Deliverable Increments agreed as "done" or only for key Deliverable Increments. Under the Fixed Price or Incremental Fixed Price pricing mechanisms, there shall be a clear linkage between an invoiced amount and the relevant Deliverable Increment.
- 3.18 In all cases there shall be a clear linkage between the Backlog and invoices submitted.
- 3.19 Invoice frequency may not align with Operational Review Events (e.g. invoicing may be monthly but review/planning events every two weeks). Under such circumstances, for clarity of receipting and audit purposes, unless agreed otherwise by the Buyer, work yet to be accepted at the Commercial Planning/Review Event shall be held back to the next appropriate invoice period.

#### Oversight of Statements of Work

- 3.20 Subsequent to the first SOW, new SOWs will be created and agreed under the concept of Call-Off Contract Management. It is anticipated that the team involved in Call-Off Contract Management shall develop the SOW requirements, work with the Supplier to agree the content of the SOW, and decide on the resourcing and pricing models prior to signing off the SOW.
- 3.21 It is the responsibility of the respective Call-Off Contract Managers to ensure SOWs are within the scope of the overarching Call-Off Contract and that budget is available under the Call-Off Contract.
- 3.22 It is envisaged that minor variations to in-place SOWs will be coordinated under the concept of Call-Off Contract Management

Providing oversight of resourcing

- 3.23 Whilst detailed planning of resource allocation to SOW tasks is handled at SOW Management level, there is an ongoing requirement to understand the total resourcing view and where resourcing is deployed at the Call-Off Contract level.
- 3.24 Under the Rainbow Team Accountability Model, the Buyer may have mandatory induction processes to be followed. The Supplier shall ensure that all such requirements are met where required prior to beginning work under any Call-Off Contract or SOW.
- 3.25 The Supplier is also responsible for ensuring that its staff (directly or indirectly employed) perform any data handover / cleansing obligations where applicable at the end of a Call-Off Contract. Call-Off Contract Management is accountable for ensuring this is done.

#### The bridge between SOW Delivery and payment / Call-Off Contract budget management

3.26 The Supplier shall, at all times, maintain an audit path linking delivery information together with invoice information.

Crown Copyright 2018

- 3.27 In an Agile environment, the contract value agreed at Call-Off Contract level is typically a maximum possible value rather than a committed sum. Subject to the terms of this Call-Off Contract, committed charges shall be set out in each SOW.
- 3.28 At all times, the Supplier shall track budgets, forecasts and actuals at purchase order level, SOW level and Call-Off Contract level and shall provide a level of granularity (e.g. monthly) as required by the Buyer.
- 3.29 The Supplier shall update such financial summaries to reflect any changes agreed under the formal Variation process.
- 3.30 The Supplier shall notify the Buyer of any forecast overspend at least 60 days in advance of such an overspend occurring.

#### **Buyer/Supplier Management**

- 3.31 Buyer/Supplier Management covers the overall relationship between Buyer and Supplier. If there is only one Call-Off Contract in place between the Buyer and Supplier the below activities may be combined under Call-Off Contract Management. However, Buyer/Supplier Management contains obligations which extend beyond specific Call-Off Contracts as set out below.
- 3.32 Buyer/Supplier Management is considered to formally start on the commencement of the Call-Off Contract. However, it is anticipated that Suppliers start to engage in Buyer/Supplier Management on an informal basis as they first become involved in a Further Competition Procedure.
- 3.33 From the Call-Off Contract Start Date the Supplier shall nominate a Buyer Account Manager and the Buyer shall nominate a Supplier Manager
- 3.34 It is expected, as part of non-Call-Off Contract-specific Supplier engagement, that the Supplier makes efforts to "understand" the Buyer. In order to improve competition and ensure a level playing the Buyer shall support attempts by the Supplier to "understand" the Buyer, without prejudicing any Further Competition Procedure.
- 3.35 It is recommended that, where a Supplier has not previously had a relationship with the Buyer, or specific relevant parts of the Buyer's business, that "get to know events" are facilitated by the Buyer.
- 3.36 Buyer/Supplier Management is considered to end on the End Date of the last effective Call-Off Contract between the Buyer and Supplier.
- 3.37 Whilst it is recognised that Buyer/Supplier Management may be incorporated within Call-Off Contract Management, there is certain information which may be aggregated up or is pertinent to the relationship with the Supplier. Examples of such information are listed under Annex 3.
- 3.38 On an ongoing basis, the Buyer and Supplier shall collaborate to ensure appropriate prioritisation of resources, focus and continuous improvement across all Call-Off Contracts between the Buyer and the Supplier.
- 3.39 Buyer SM Events, for handling the above matters, shall be held at a frequency determined by the Buyer.

#### Collaborative Buyer/Supplier Events

- 3.40 Periodically, the Buyer may elect to meet with all Suppliers with whom it has Call-Off Contracts or who are actively participating in Framework business with the Buyer.
- 3.41 Typically such events will be held quarterly, but the frequency shall be determined by the Buyer. Suppliers should make every reasonable effort to participate in, and contribute to, such events.
- 3.42 The purpose of such events will typically be to:
  - 3.42.1 provide the Buyer with the opportunity to share, on an equitable basis, future pipelines of work;
  - 3.42.2 provide the Buyer with the opportunity to share future technology trends from the Buyer's perspective;
  - 3.42.3 provide Suppliers with the opportunity to suggest overall improvements to the way the Framework Contract is working with the Buyer;
  - 3.42.4 share emerging technologies coming out of Supplier activities;
  - 3.42.5 identify where there may be market shortages in skills and discuss mechanisms (training, knowledge sharing, buddying/mentoring, etc.) to address such shortages;
  - 3.42.6 agree proposals for possible changes to be considered by CCS;
  - 3.42.7 discuss opportunities to improve the operation of the overall Framework Contract;
  - 3.42.8 discuss proposed updates to the Framework Contract; and
  - 3.42.9 any other activity which may be proposed by the Buyer.

#### 4. KEY MECHANISMS

- 4.1 Product Backlog (Scrum®) or Prioritised Requirements List (AgilePM®) or equivalent:
  - 4.1.1 Typical information to be held in the Backlog may be found in Annex 4;
  - 4.1.2 The Backlog should be considered to be the operational equivalent of a change control log capturing refinements, changes, additions and deletions. The Backlog demonstrates the value delivered (even if only at Deliverable Increment) and provides an indicator on how much change the project/programme has had to absorb;
  - 4.1.3 The Backlog evidences value delivered; and
  - 4.1.4 Since the Backlog captures ongoing refinement, Variations should only be required to changes to funding, scope and high level Milestones/timescales.

#### 4.2 Variation and Change Management

4.2.1 The term "SOW Variation" is used to describe changes to budget, timescales, and scope at the SOW level. Other than the Buyer ensuring finances are still available and that the scope of the SOW remains inside the Call-Off Contract, such SOW Variations are intended to be managed

> within the Call-Off Contract. As such extended Buyer governance is not anticipated.

- 4.2.2 The term "Variation", in this context, is used to describe changes to overall contract value, overall contract milestones and/or term and where overall scope requires to change and such Variations shall follow the procedure set out at Clause 24.
- 4.2.3 The Supplier is required to maintain copies of all Variation and SOW Variation documentation.

#### 4.3 Balanced Scorecard

- 4.3.1 Where the Buyer determines, the Supplier shall work with the Buyer to develop the detail of a Balanced Scorecard.
- 4.3.2 The Buyer shall give notice to the Supplier as to when the Balanced Scorecard shall become effective.
- 4.3.3 The principles outlined in Procurement Policy Note 09/16: Procurement for Growth Balanced Scorecard (or any later replacement) shall apply.
- 4.3.4 The Parties shall refer to the Balanced Scorecard Paper, Annex A: Diagram and Annex B: Strategic Themes and Critical Success Factors associated with Procurement Policy Note 09/16 (or latest equivalents) when formulating a Balanced Scorecard.
- 4.3.5 The Buyer may elect to apply a Balanced Scorecard either per Call-Off Contract or at Supplier level.
- 4.3.6 The frequency of update to the Balanced Scorecard shall be determined by the Buyer (but will generally align with Commercial Planning/Review Events at Call-Off level and/or Buyer SM Events at Supplier level).
- 4.3.7 An example of a Balanced Scorecard may be found in Annex 5.

#### 5. RISK MANAGEMENT

- Reference is made to the HM Treasury Green Book supplementary guidance on risk (https://www.gov.uk/government/publications/green-book-supplementary-guidancerisk).
- 5.2 In particular, the principles and concepts contained in the Orange Book shall underpin the risk management practices implemented under this Schedule.
- 5.3 Both Parties shall pro-actively manage risks attributed to them under the terms of this Call-Off Contract.
- 5.4 The Buyer will aim to ensure that the placement of risk is appropriate (i.e. risk is placed where it can be influenced).
- 5.5 The Supplier shall develop, operate, maintain and amend, as agreed with the Buyer, processes for:
  - 5.5.1 the identification and management of risks;
  - 5.5.2 the identification and management of issues; and

- 5.5.3 monitoring and controlling project plans.
- 5.6 The Supplier shall allow the Buyer to inspect at any time within the Supplier's working hours the accounts and records which the Supplier is required to keep.
- 5.7 The Supplier will maintain a risk register of the risks relating to the Call-Off Contract which the Buyer and the Supplier have identified.
- 5.8 The Buyer may elect, at any point in time, to conduct ongoing Supplier risk management as follows:
  - 5.8.1 the Buyer may choose to use credit rating checks (such as those offered by Experian and Dun & Bradstreet) to monitor the financial health of the Supplier;
  - 5.8.2 should the Buyer determine that a Supplier could be at financial risk, the Buyer may request financial details (including current unpublished accounts) in order to better understand any risk which could have an impact on the Call-Off Contract;
  - 5.8.3 on request by the Buyer, the Supplier shall provide and work to a financial risk mitigation plan as a means of protecting the interests of the Buyer; and
  - 5.8.4 the Supplier shall take reasonable steps to ensure the financial health of any Subcontractors it engages with. In the event of a potential risk with any Subcontractor the Supplier shall notify the Buyer of such risks and the mitigation actions it is taking to protect the interests of the Buyer.

#### 6. KEY ROLES

- 6.1 Key Roles and Key Staff are identified and named in each SOW.
- 6.2 The Supplier and the Buyer shall each nominate an Operational Contract Manager for the purposes of this Call-Off Contract through whom the provision of the Deliverables shall be managed day-to-day.
- 6.3 The Supplier and the Buyer shall each nominate a Call-Off Contract Manager for the purposes of this Call-Off Contract through whom commercial matters may be escalated as and when needed and at a regular frequency as determined by the Parties.
- 6.4 The Parties shall ensure that appropriate resource is made available on a regular basis such that the contract management aims, objectives and specific provisions of this Call-Off Contract can be fully realised.

#### **Operational Contract Managers**

- 6.5 Operational contract management shall cover matters including:
  - 6.5.1 developing technical scope for individual SOWs;
  - 6.5.2 ongoing joint management of Backlog item lists;
  - 6.5.3 resource monitoring; and
  - 6.5.4 progress against Deliverables and reporting and receipting of the same.
- 6.6 The Supplier's Operational Contract Manager shall be:

> 6.6.1 the primary point of contact to receive operational communications from the Buyer and will also be the person primarily responsible for providing operational information to the Buyer;

- 6.6.2 able to delegate his position to another person at the Supplier but must inform the Buyer before proceeding with the delegation and it will be delegated person's responsibility to fulfil the Operational Contract Manager's responsibilities and obligations;
- 6.6.3 able to cancel any delegation and recommence the position himself; and
- 6.6.4 replaced only after the Buyer has received notification of the proposed change.
- 6.7 The Buyer's Operational Contract Manager is the Buyer side of the Operational Contract Management relationship, providing operational communications to the Supplier, acknowledging receipt of Deliverables, and having equivalent obligations with respect to delegation and notification of any changes.
- 6.8 The Buyer may provide revised instructions to the Supplier's Operational Contract Manager in regards to the operation of the Call-Off Contract and it will be the Supplier's Operational Contract Manager's responsibility to ensure the information is provided to the Supplier and the actions implemented.
- 6.9 Receipt of communication from the Supplier's Operational Contract Manager by the Buyer does not absolve the Supplier from its responsibilities, obligations or liabilities under the Call-Off Contract.

#### Call-Off Contract Managers

- 6.10 Commercial Contract Management shall cover matters including:
  - 6.10.1 development of commercial scope for individual SOWs;
  - 6.10.2 ensuring compliance with Call-Off Contract terms;
  - 6.10.3 oversight of commercial performance of the Call-Off Contract; and
  - 6.10.4 resolution of commercial issues, including any need for contractual dispute resolution or escalation.
- 6.11 The Supplier's Call-Off Contract Manager shall be:
  - 6.11.1 the primary point of contact to receive commercial communications from the Buyer and will also be the person primarily responsible for providing commercial information to the Buyer;
  - 6.11.2 able to delegate his position to another person at the Supplier but must inform the Buyer before proceeding with the delegation and it will be delegated person's responsibility to fulfil the Call-Off Contract Manager's responsibilities and obligations;
  - 6.11.3 able to cancel any delegation and recommence the position himself; and
  - 6.11.4 replaced only after the Buyer has received notification of the proposed change.

Crown Copyright 2018

- 6.12 The Buyer's Call-Off Contract Manager is the Buyer side of the Commercial Contract Management relationship, providing commercial communications to the Supplier and having equivalent obligations with respect to delegation and notification of any changes.
- 6.13 The Buyer may provide revised instructions to the Supplier's Call-Off Contract Manager in regards to the commercial aspects of the Call-Off Contract and it will be the Supplier's Call-Off Contract Manager's responsibility to ensure the information is provided to the Supplier and the actions implemented.
- 6.14 Receipt of communication from the Supplier's Call-Off Contract Manager by the Buyer does not absolve the Supplier from its responsibilities, obligations or liabilities under the Call-Off Contract.

#### Supplier Manager and Account/Buyer Manager

- 6.15 The Buyer side Supplier Manager, if named by the Buyer, is the individual with oversight of the Supplier as a whole and shall be the prime contact for the Supplier's Account/Buyer Manager.
- 6.16 Buyer matters which are not able to be dealt with at the Call-Off Contract level shall be escalated to the Supplier Manager.
- 6.17 The role shall delegate to the Buyer's Call-Off Contract Manager if not named.
- 6.18 The Supplier side Account/Buyer Manager, if named by the Supplier, is the individual with the relationship with the Buyer as a whole and shall be the prime contact for the Buyer's Supplier Manager.
- 6.19 Supplier matters which are not able to be dealt with at the Call-Off Contract level shall be escalated to the Supplier's Account/Buyer Manager.
- 6.20 The role shall delegate to the Supplier's Call-Off Contract Manager if not named.

#### **Contract Boards**

- 6.21 One or more Contract Boards may, at the Buyer's sole option be established for the purposes of this Call-Off Contract. At minimum the Supplier and the Buyer shall be represented on the board.
- 6.22 Where required, the board members, frequency and location of board meetings and planned start date by which the board shall be established shall be set out in Annex 6.
- 6.23 In the event that either Party wishes to replace any of its appointed board members, that Party shall notify the other in writing for approval by the other Party (such approval not to be unreasonably withheld or delayed). Each Buyer board member shall have at all times a counterpart Supplier board member of equivalent seniority and expertise.
- 6.24 Each Party shall ensure that its board members shall make all reasonable efforts to attend board meetings at which that board member's attendance is required. If any board member is not able to attend a board meeting, that person shall use all reasonable endeavours to ensure that a delegate attends the Operational Board meeting in his/her place (wherever possible) and that the delegate is properly briefed and prepared and that he/she is debriefed by such delegate after the board meeting.

6.25 The purpose of the board meetings will generally be to review the performance of the Call-Off Contract. The agenda for each meeting shall be set by the Buyer and communicated to the Supplier in advance of that meeting.

#### Executive Sponsors

6.26 Each party may elect to nominate an Executive Sponsor for the purposes of ensuring corporate alignment with the overall Buyer/Supplier relationship as well as acting as a point of escalation to assist in removing potential corporate blockers.

#### 7. KEY INFORMATION

- 7.1 The Buyer requires the Supplier to provide the management information as identified in the Annexes to this Schedule. These requirements are without limitation to the Buyer's right to require the submission of information, reports, records and data as set out elsewhere in the Call-Off Contract.
- 7.2 The Supplier shall, within 30 days of the earliest of:
  - 7.2.1 the date which is 30 days before the end of the Call-Off Contract Period;
  - 7.2.2 receipt of a Termination Notice;
  - 7.2.3 notification by the Buyer of an actual or intended Service Transfer; or
  - 7.2.4 a written request by the Buyer,

provide the Buyer with a complete set of up to date information in respect of all types of information set out in the Annexes.

#### 8. INVOICING AND PAYMENT

- 8.1 The frequency of invoicing shall be on a Monthly basis, unless agreed by the Buyer, in writing, as otherwise.
- 8.2 Invoices for the preceding Month shall be submitted within 10 Working Days of the end of the Month unless agreed, in writing, by the Buyer as otherwise.
- 8.3 Separate invoices shall be provided for each SOW, clearly identifying the Call-Off Contract and SOW which the invoice relates to.
- 8.4 The information to be provided by the Supplier with each Invoice shall be that identified in Annex 7.
- 8.5 In order to facilitate prompt payment of invoices, the Supplier shall coordinate with the Buyer to ensure acceptance of Deliverables completed.
- 8.6 The Buyer shall notify the Supplier of any incorrect Invoice submissions within 5 Working Days of receipt. Incorrect invoices shall include (without limitation) invoices for work which has not been accepted by the Buyer and invoices based on incorrect rates. The Supplier shall resubmit invoices once corrected.
- 8.7 Unless otherwise agreed, the Buyer shall pay all accepted invoices within 30 days of submission.

#### 9. MATERIAL KPIs

- 9.1 The Supplier's performance shall be measured by the Material KPIs set out in the Order Form.
- 9.2 The Supplier shall comply with the Material KPIs and establish processes to monitor its performance against them and the Supplier's achievement of Material KPIs shall be reviewed during Commercial Planning/Review Events.
- 9.3 The Buyer reserves the right to adjust, introduce new, or remove Material KPIs throughout the Call-Off Contract Period, however any significant changes to Material KPIs shall be agreed between the Buyer and the Supplier in accordance with the Variation Procedure.
- 9.4 The Buyer reserves the right to use and publish the performance of the Supplier against the Material KPIs without restriction.
- 9.5 In the event that the Buyer and the Supplier are unable to agree the performance score for any Material KPI during a Commercial Planning/Review Event, the disputed score shall be recorded and the matter shall be referred to the Buyer Authorised Representative and the Supplier Authorised Representative in order to determine the best course of action to resolve the matter (which may involve organising an ad-hoc meeting to discuss the performance issue specifically).
- 9.6 In cases where the Buyer Authorised Representative and the Supplier Authorised Representative fail to reach a solution within a reasonable period of time, the matter shall be referred to the Dispute Resolution Procedure.
- 9.7 Failures to meet Material KPIs shall be addressed in accordance with Paragraphs 5.4 and 5.5 of Framework Schedule 4 (Framework Management).

# Annex 1: SOW Management Related Information

Topic Frequency or Details		
Operational Planning Event Frequency	Fortnightly	
Operational Review Event Frequency	Fortnightly	

### I

The following table provides a list of information required by the Buyer as part of the SOW Management process.

#### Notes:

- There are some artifacts which held at Call-Off Management level which could equally apply at this level. Please read that list too (Annex 2).
- Supplier and Subcontractor related topics are covered under Buyer/Supplier Management
- 3. This list is non-exclusive, and the Buyer may elect to include other information.

	Type of Information	Required ?	Refresh Frequency
A1.01	Backlog (Task level)	Yes	Operational Planning Event
A1.02	Forecast Resource Profile	Yes	Operational Planning Event
A1.03	Actual Resource Utilisation	Yes	Operational Review Event
A1.04	Task Completion Status	Yes	Operational Review Event
1	Z XX XX	~	
7	1		
Typical	Anile Artifacts (Statement of Work Information	1	
Typical A1.10	Agile Artifacts (Statement of Work Information Sprint Burndown Report	) Y	Supplied via Jira
A1.10	Sprint Burndown Report	Y	Jira Supplied via
A1.10 A1.11	Sprint Burndown Report Velocity Chart	Y Y	Jira Supplied via Jira Supplied via
A1.10 A1.11 A1.12	Sprint Burndown Report Velocity Chart Epic and Release Burndown Chart	Y Y Y	Jira Supplied via Jira Supplied via Jira
A1.10 A1.11 A1.12 A1.13	Sprint Burndown Report Velocity Chart Epic and Release Burndown Chart Agile Control Chart	Y Y Y N	Jira Supplied via Jira Supplied via Jira [N/A]

# Annex 2: Call-Off Contract Management Information

Frequency or Details
Monthly]
1

The following table provides a list of information required by the Buyer as part of the Call-Off Contract Management process (note that this is a logical model and the Buyer may elect to operate "events" under different labels).

#### Notes:

- 1. This list is non-exclusive, and the Buyer may elect to include other information.
- This list is also illustrative and non-exhaustive. It is anticipated the Buyer will review and edit these tables as part of preparing a Call-Off Contract (and reserves the right to amend during the Call-Off Contract Period).

Ref.	Type of Information	Required ?	Refresh Frequency
Contra	ct Management Artifacts		
A2.01	Backlog (Deliverable Increment Level) (including agreements to Acceptance, Pricing, Status, etc – see Backlog for details)	Yes	Commercial Planning / Review Event
A2.02	List of Supplier Staff including: SOW, Unique ID, Name, Job Role, link to Call-Off Rate Card, rate, Start Date, End Date, Planned Days, Forecast Days, Actual Days, Security Level, Contractor Status, IR35 determination, etc as specified by the Buyer)	Yes	Commercial Planning / Review Event
A2.03	Cost/Spend Cash Flow Data including: SOW, Budget, Forecast, Actual, links to POs, Variations, etc as specified by the Buyer	Yes	Commercial Planning / Review Event
A2.04	Risks, Issues (and Opportunities) Log including impact assessment, mitigation details, etc.	Yes	Commercial Planning / Review Event
A2.05	Any remediation plans agreed, and any progress against an agreed remediation plan.	Yes	Commercial Planning / Review Event
A2.06	Statement of any commercial issues by the Supplier during that period (late payment). Report of resolution of any previously noted items.	Yes	Commercial Planning / Review Event
A2.07	Copies of notices received and/or issues by the Buyer, and referrals to any Dispute Resolution Procedure	Yes	On demand by the Buyer
A2.08	Variations detailing material changes to scope, cost, major milestones and/or overall timescales	Yes	On demand by the Buyer
A2.09	Stakeholder Map	Y	To be determined following SOW 2
A2.10	Communications Plan	Y	To be determined following SOW 2
A2.11	Copies of all Contract related documents including referenced documents, relevant Sub-Contracts and any ancillary documents (including amendments)	Yes	On demand by the Buyer
A2.12	Current contact and address for notices under the Call- Off Contract.	Yes	Immediate and updates following any changes

Ref.	Type of Information	Required ?	Refresh Frequency
	[Please insert any other specific details relating to the provision of management information required by the Buyer.]	N	[N/A]

Ref.	Type of Information	Required ?	Refresh Frequency
Typica Consor	I AgilePM® Products (ref: Agile Project Management Ha tium): https://www.agilebusiness.org/page/ProjectFrame	Stores and and the	
A2.20	Terms of Reference	N	[N/A]
A2.21	Business Case	N	[N/A]
A2.22	Solution Architecture Definition	Y	[To be determined following SOW 2]
A2.23	Development Approach Definition	Y	To be determined following SOW 2]
A2.24	Delivery Plan (also under Implementation Plans)	Y	[To be determined following SOW 2]
A2.25	Timebox Plans (also under Implementation Plans)	Y	[To be determined following SOW 2]
A2.26	Timebox Review Records	Y	[To be determined following SOW 2]
A2.27	Management Approach Definition	Y	[To be determined following SOW 2]
A2.28	Feasibility Assessment	Y	To be determined following SOW 2]
A2.29	Foundations Summary	Y	[To be determined following SOW 2]
A2.30	Project Review Report	Y	[To be determined following SOW 2]
4	[Please insert any other specific details relating to the provision of management information required by the Buyer.]	N	[N/A]
Other	Agile Products (courtesy Blended Agile Delivery www.the	badtoolkit.com	)
A2.40	VMOST Mission Boards	N	[N/A]
A2.41	ROI Projections	N	[[N/A]
A2.42	EPIC Board (also under Implementation Plans)	Y	[To be determined following SOW 2]
A2.43	User Stories	Y	[To be determined following SOW 2]
A2.44	Release Propositions	Y	[To be determined following SOW 2]
A2.45	Proof of Concepts (POCs) / Spikes	Y	[To be determined following SOW 2]
A2.46	Designs / UX (User Functionality)	Y	[To be determined following SOW 2
A2.47	Business Architecture Changes	Y	[To be determined following SOW 2]
A2.48	Data Changes	Y	[To be determined following SOW 2]

Ref.	Type of Information	Required ?	Refresh Frequency
A2.49	Candidate Architectures	Y	[To be determined following SOW 2]
A2.50	PTEST Requirements	N	[N/A]
A2.51	Elaboration Boards (also under Implementation Plans)	N	[N/A]
A2.52	Spring Boards (also under Implementation Plans)	N	[N/A]
A2.53	Delivery Metrics	Y	[To be determined following SOW 2]
A2.54	Health Checks	Y	[To be determined following SOW 2]
	[Please insert any other specific details relating to the provision of management information required by the Buyer.]	N	[N/A]

Ref.	Type of Information	Required ?	Refresh Frequency
Contrit	outions to Other Programme / Project Management Arti	facts (not liste	d above)
A2.60	Technical Requirements (Functional / Non-Functional)	Y	[To be determined following SOW 2]
A2.61	Technical Constraints (e.g. compatibility with existing systems)	Y	[To be determined following SOW 2]
A2.62	Pre-procurement support activities	N	[N/A]
A2.63	Make or Buy Analysis	N	[N/A]
A2.64	Technical Stack Specification	Y	[To be determined following SOW 2]
A2.65	Technical Road Map	Y	To be determined following SOW 2]
A2.66	Target Operating Model	Y	[To be determined following SOW 2]
A2.67	Skills Requirements Profile	Y	[To be determined following SOW 2]
	[Please insert any other specific details relating to the provision of management information required by the Buyer.]	N	N/A]

Ref.	Type of Information	Required ?	Refresh Frequency
Securi	ty and Personal Data Reporting Information		
A2.70	DSP Toolkit Submissions	Yes	Annually in arrears
A2.71	Copies of required security clearance certificates for all staff (where staff process Patient Identifiable Data or are on Buyer Premises or access Buyer Systems)	Yes	[On demand by the Buyer]
A2.72	Unique individual identifier number and full name of staff handling Patient Identifiable Data (where staff process Patient Identifiable Data or are on Buyer Premises or access Buyer Systems)	Yes	[On demand by the Buyer]
A2.73	Updated DPIA	No	N/A
A2.74	Completed Information Security Management (ISM)Document Set	No	N/A

Ref.	Type of Information	Required ?	Refresh Frequency
A2.75	Evidence of data destruction certification	Yes	On demand by the Buyer
A2.76	Notices of any breach of data provision or security provisions	Yes	On occurrence

Ref.	Type of Information	Required ?	Refresh Frequency
Hando	ver / Exit Related Information		
A2.80	Records required by Law as they relate to the provision of the services (including in relation to health and safety matters and health and safety files), such records to be prepared in accordance with any applicable laws or regulations.	Yes	On demand by the Buyer
A2.81	All training, implementation, operation and maintenance manuals related to the provision of the Deliverables.	Y	On demand by the Buyer
A2.82	All certificates, licences, registrations or warranties related to the provision of Deliverables	Y	On demand by the Buyer
A2.83	Exit Plan (and any requested updates)	Y	(see Call-Off Schedule 10: Exit Management)
	[Please insert any other specific details relating to the provision of management information required by the Buyer.]	N	[N/A]

Ref.	Type of Information	Required ?	Refresh Frequency
TUPE	nformation		
A2.90	Details of service functions that have provided services to the Buyer, and the denoting characteristics that delineate the functions (including but not limited to function code references and names used on the Supplier's finance system, and period of time for which such code has existed).	N	On demand by the Buyer
A2.91	Details of the number of customers supported by the Supplier's named service functions.	N	On demand by the Buyer
A2.92	The Staffing Information (as defined in Call-Off Schedule 2 (Staff Transfer) Buyer in relation to an anticipated potential Service Transfer or as required to be provided under the Schedule	N	On demand by the Buyer
	[Please insert any other specific details relating to the provision of management information required by the Buyer.]	N	[N/A]

# Annex 3: Buyer/Supplier Management Information

Topic	Frequency or Details	
Buyer SM Event Frequency	[Quarterly]	
buyer Sin Event requency	[Quarterly]	

The following table provides a list of information required by the Buyer as part of the Buyer/Supplier Management process (note that this is a logical model and the Buyer may elect to operate "events" under different labels).

#### Notes:

- 1. This list is non-exclusive, and the Buyer may elect to include other information.
- This list is also illustrative and non-exhaustive. It is anticipated the Buyer will review and edit these tables as part of preparing a Call-Off Contract (and reserves the right to amend during the Call-Off Contract Period).
- Information which could be seen to be advantageous to all Suppliers or which would be of benefit to all parties are dealt with under Buyer Specific Framework Management.

Ref.	Type of Information	Required ?	Refresh Frequency
Suppli	er and Sub-Contract Related Information		
A3.01	Report from the Supplier of any change to its financial standing / any change to its group structure.	Yes	On demand by the Buyer
A3.02	Request for use of any additional Subcontractors not forming part of the original Framework either: As part of a proposed response to a new Further Competition Procedure; or During execution of an existing Call-Off Contract	Yes	Prior to competition. Prior to mobilisation (as appropriate)
A3.03	Written confirmation by the Supplier, confirmed in writing by the Subcontractor(s), that they have in a place a contract which mirrors the provisions of the Call-Off Contract with the Supplier	Yes	On demand by the Buyer
A3.04	Written assurance by any Subcontractor that the provisions under A3.03 are also cascaded down their supply chain	Yes	On demand by the Buyer
A3.05	Copies of published financial accounts	Yes	On demand by the Buyer
A3.06	In the event of the Buyer becoming aware of any financial difficulties being faced by the Supplier (as reasonably judged by the Buyer), copies of current accounts (whether published or not)	Yes	On demand by the Buyer
	[Please insert any other specific details relating to the provision of management information required by the Buyer.]	N	[N/A]

Ref.	Type of Information	Required ?	Refresh Frequency
Aaarea	ated Views		
A3.10	Summary of all Call-Off Contracts, Variations and SOWs from a performance perspective (possibly in the form of a traffic light report). All as required by the Buyer	Yes	Buyer SM Event
A3.11	Summary of cash flow (budget, forecast, actual) for all Call-Off Contracts and SOWs	Yes	Buyer SM Event
A3.12	Summary of resources deployed and any people issues requiring resolution	Yes	Buyer SM Event
A3.13	Summary of escalated risks, issues and other escalation topics (e.g. around obligations)	Yes	Buyer SM Event
A3.14	Quality Review Heat Map (or equivalent) capturing an assessment of quality against things like User needs, roadmaps, technology, delivery, transparency, decision making, go-live readiness, etc as defined by the Buyer	Yes	Buyer SM Event
	[Please insert any other specific details relating to the provision of management information required by the Buyer.]	N	N/A]
Suppli	er (and Buyer) Performance and Continuous Improveme	nt	
A3.20	Overall "Temperature Checks" or "Office Vibe" status or equivalent – providing 360° feedback on things like relationships, recognition, growth and development, wellness, happiness, etc (May also be at individual Call-Off Contract level)	Y	Buyer SM Event
A3.21	Balanced Scorecard (as defined in detail in paragraph 4.3)	Y	Buyer SM Event
A3.22	Mutual personnel development opportunities	Y	As mutually agreed
A3.23	Continuous improvement / opportunities	Y	As mutually agreed
<	[Please insert any other specific details relating to the provision of management information required by the Buyer.]	N	[N/A]
Overal	Supplier Governance		
A3.30	Summary of agreed remediation actions and their status	Yes	Buyer SM Event
A3.31	Mutual Review and endorsement of Call-Off Contract Variations	Yes	On demand by the Buyer
A3.32	Recommendations to be taken to framework level (under the management of the Authority)	Yes	As mutually agreed
	[Please insert any other specific details relating to the provision of management information required by the Buyer.]	N	N/A]

# Annex 4: Product Backlog/Prioritised Requirements List

Please refer to 'Annex 4: Call-Off Schedule 20 – Product Backlog Item List', within 'Call-Off Schedule 20 (Call-Off Specification)'.

Framework Ref: RM6221 Project Version: v1.0 Model Version: v3.1

# Annex 5: Balanced Scorecard (Example)

 In addition to the Supplier's performance management obligations set out in the Framework Contract, the Buyer and the Supplier may agree to the following Balanced Scorecard & KPIs for this Call-Off Contract (see Balanced Scorecard examples below). However, the Balanced Scorecards may change and be amended over the life of the Contract.

Example 1

	variipie conc	ract Scorecard					
	al-sector Market	Stagend Autoget	Annual	Radal Same	1 1		
Carry 9	sparsted Barris	ni	annulus.	anist Street	-		
hites	ed Balance Bestudies	and a second sec		-			
		forest Scenet Some	71.6	1000			
		1 Part Andrew March and Ma		71.6	8		
		2 guardent diright fit		P			
		T Total and the Park No. 19	The Transmission lines	Material	(Antoi apt 8		
-	KPis	· COMPANY PLANTER NAME	E Vennes Areas	E Vonante Arabian 🖉 Bras Balanter B			
			N The				
		Parlamenta Maria	Art Barger Arts Sugarant	danie -	-		
43	Indep fire Free Annuel (	14)					
ŧ.	this looks by payments a	5m	175	- 10			
•	Average and the second	N.		- 6	+		
6	Propriet Stephen (and that the c	las .		- 10			
4	Real Sector In Sector (1)	6.00.00	94.				
۲.	All families and	nit.wr	-		+		
	Rest when which have on	N	-	- 17			
۳.							

Example 2

C52340 NHS Pathways 03 Call Off Schedules 1.0F 22Dec21

#### Call-Off Schedule 15A (Health Supplier and Contract Management) Call-Off Ref: Crown Copyright 2018



# Example 3

and balances of first 1977				1.1
				100
or Radiation Performance Damper	a deserve			
Research ( Despite	date:	ines (		1 (e)
Para.				
	-		190	y 71
and the second s	100			
Contract of the local				
Carlos Ca				-
17777 C		-		
and Participants Talkans				Barrison .
and the second se				
2.1			122	
1.2				

### Example 4



- The purpose of the Balanced Scorecard is to promote contract management activity, through measurement of a Supplier's performance against key performance indicators, which the Buyer and Supplier should agree at the beginning of a Call-Off Contract. The targets and measures listed in the example scorecard (above) are for guidance and should be changed to meet the agreed needs of the Buyer and Supplier.
- 3. The recommended process for using the Balanced Scorecard is as follows:
  - The Buyer and Supplier agree a templated Balanced Scorecard together with a
    performance management plan, which clearly outlines the responsibilities and
    actions that will be taken if agreed performance levels are not achieved.
  - On a pre-agreed schedule (e.g. monthly), both the Buyer and the Supplier provide a rating on the Supplier's performance
  - Following the initial rating, both Parties meet to review the scores and agree an overall final score for each key performance indicator
  - Following agreement of final scores, the process is repeated as per the agreed schedule.

# Annex 6: Contract Boards

To be confirmed prior to contract execution.

If required, the Parties agree to operate the following boards at the locations and at the frequencies set out below:

Meeting	Purpose	Location	Frequency
1.			
2.			
3.			
4.			
1. 2. 3. 4. 5.			

Buyer	March Process	
Operational Contract Manager		
Commercial Lead	[insert name]	
[Title 3]	[insert name]	
Supplier		
[Title 1]	[insert name]	
[Title 2]	[insert name]	
Other		
[Title 1]	[insert name]	
[Title 2]	[insert name]	
		_

# Annex 7: Information to be provided on Invoice

The following information is required the be provided along with each invoice:

Ref.	Type of Information	Which Services does this requirement apply to?	Required regularity of Submission
1.1	Details of the reference for the Call-Off Contract.	All	with each invoice
1.2	Details of the reference of the Statement of Work. To include reference to Milestones.	All	with each invoice
1.3	Details of the reference of the Variation.	All	with each invoice
1.4	Unique invoice number.	All	with each invoice
1.5	Buyer Purchase Order number (allocated on a per Programme basis by the Buyer).	All	with each invoice
1.6	Date of invoice.	All	with each invoice
1.7	Portfolio reference and programme name.	All	with each invoice
1.8	The period(s) to which the relevant Charge(s) relate.	All	with each invoice
1.9	Details of payments due in respect of achievement of a milestone.	All	with each invoice
1.10	Details of any service credits or similar incentives that shall apply to the charges detailed on the invoice.	All	with each invoice
1.11	The total charges gross and net of any applicable All deductions and, separately, the amount of any disbursements properly chargeable to the Buyer under the terms of this Call-Off Contract, and separately.		with each invoice
1.12	Any VAT or other sales tax payable in respect of the same.	All	with each invoice
1.13	A contact name and telephone number of a All responsible person in the Supplier's finance department in the event of administrative gueries.		with each invoice
1.14	The banking details for payment to the Supplier via All electronic transfer of funds (i.e. name and address of		with each invoice
1.15	bank, sort code, account name and number).         5       Detailed time sheets for any time and materials-based pricing, including unique individual identifier number (and same number to be used in all correspondence, reports, provision of information etc. in relation to Supplier's staff), and full name. To include description of type of work undertaken, role and Team description.       All		with each invoice
1.16	Copies of invoices including VAT information for any expenses-based disbursements and deductions.	All	with each invoice
1.17	Asset number/ Asset type, (hardware/software, perpetual licence or subscription) description and period covered of any purchased / licensed / leased items.	Asset type, (hardware/software, All with each or subscription) description and	
1.18	Where appropriate, details of user stories worked on by each individual.	Software Development services	with each invoice

The information below offers an example of how various types of charges might be best detailed on an invoice:

Buyer Details Buyer Details Buyer Details						Supplier D Supplier D Supplier D	Details	
Buyer Details						Supplier D		
							Invoice # Invoice Date Purchase Order	10000 102 <sup>1</sup> 07/10 10000000
Portfolio Ref (P	Doorbox)	Programme Name						
SOW KKK	Variation (CCNo	() SOW Title						
Mestone		a				Period	800X	
	Reference #	Description of Work completed						Amount
	Reference #	Description of Work completed						Amount
Contraction and the	Reference #	Description of Work completed						Amount
Deliverable	Reference #	Description of Work completed						Anioun
Team	Description					Period	1000	
	Role	unique individual identifier #	Type of Work (e.g.Live Service, Development,	Decement		Day Rale		Amoun
	Role							Amoun
	Role	unique individual identifier # unique individual identifier #	Type of Work (e.g.Live Service, Development, Type of Work (e.g.Live Service, Development,			Day Rate Day Rate		Anour
	Role							
		unique individual identifier #	Type of Work (e.g.Live Service, Development,			Day Rate		Anour
Name	Role	unique individual identifier #	Type of Work (e.g. Live Service, Development,	Discovery)		Day Rate	Days	Amoun
Hardware						Period	800	
Product	Product Code	Serial #	Type (e.p. Server, Laptop, Tablet, Mobile Pho-	rieti				Amoun
Product	Product Code	Serial #	Type (e.p. Server, Laptop, Tablet, Mobile Pho-					Amount
Product	Product Code	Serial #	Type (e.g. Server, Laptop, Tablet, Mobile Pho					Anoun
Software Licent						Period	8003	
	Description :	Perpetual/Subscription	Lines	ce Period	From	To	# Users	Anoun
	Description	Perpetual/Subscription		ce Period	From	To	# Users	Amoun
	Description			ce Period	From	To	# Users	Amoun
Hodect	Description	Perpetual/Subscription	Losin	ce menos	From	10	# Users	Amoun
Expenses						Period	1000	
Name	Date Incurred	Type (Travel, Accomodation etc	Description including Provider					Amount
Name	Date Incurred	Type (Travel, Accomodation etc						Anout
Credita					Pe	riod Applied	1000	
Description						Period Re		Anour
						1000000		
							Net Total VAT / Sales Tax	EXOC EXOC
							Gross Total	DOO
Banking Details	100							
Account Name								
Dank:								
Sort Code:			Supplier Finance Department					
Account No: BAN Code:			Contact Name: Contact Number:					

# Call-Off Schedule 16 (Benchmarking)

# 1. DEFINITIONS

1.1 In this Schedule, the following expressions shall have the following meanings:

meanings.	
"Benchmark Review"	a review of the Deliverables carried out in accordance with this Schedule to determine whether those Deliverables represent Good Value;
"Benchmarked Deliverables"	any Deliverables included within the scope of a Benchmark Review pursuant to this Schedule;
"Comparable Rates"	the Charges for Comparable Deliverables;
"Comparable Deliverables"	deliverables that are identical or materially similar to the Benchmarked Deliverables (including in terms of scope, specification, volume and quality of performance) provided that if no identical or materially similar Deliverables exist in the market, the Supplier shall propose an approach for developing a comparable Deliverables benchmark;
"Comparison Group"	a sample group of organisations providing Comparable Deliverables which consists of organisations which are either of similar size to the Supplier or which are similarly structured in terms of their business and their service offering so as to be fair comparators with the Supplier or which, are best practice organisations;
"Equivalent Data"	data derived from an analysis of the Comparable Rates and/or the Comparable Deliverables (as applicable) provided by the Comparison Group;
"Good Value"	that the Benchmarked Rates are within the

Upper Quartile; and

### "Upper Quartile"

in respect of Benchmarked Rates, that based on an analysis of Equivalent Data, the Benchmarked Rates, as compared to the range of prices for Comparable Deliverables, are within the top 25% in terms of best value for money for the recipients of Comparable Deliverables.

### 2. When you should use this Schedule

- 2.1 The Supplier acknowledges that the Buyer wishes to ensure that the Deliverables, represent value for money to the taxpayer throughout the Contract Period.
- 2.2 This Schedule sets to ensure the Contracts represent value for money throughout and that the Buyer may terminate the Contract by issuing a Termination Notice to the Supplier if the Supplier refuses or fails to comply with its obligations as set out in Paragraphs 3 of this Schedule.
- Amounts payable under this Schedule shall not fall with the definition of a Cost.

### 3. Benchmarking

### 3.1 How benchmarking works

- 3.1.1 The Buyer and the Supplier recognise that, where specified in Framework Schedule 4 (Framework Management), the Buyer may give CCS the right to enforce the Buyer's rights under this Schedule.
- 3.1.2 The Buyer may, by written notice to the Supplier, require a Benchmark Review of any or all of the Deliverables.
- 3.1.3 The Buyer shall not be entitled to request a Benchmark Review during the first six (6) Month period from the Contract Commencement Date or at intervals of less than twelve (12) Months after any previous Benchmark Review.
- 3.1.4 The purpose of a Benchmark Review will be to establish whether the Benchmarked Deliverables are, individually and/or as a whole, Good Value.
- 3.1.5 The Deliverables that are to be the Benchmarked Deliverables will be identified by the Buyer in writing.
- 3.1.6 Upon its request for a Benchmark Review the Buyer shall nominate a benchmarker. The Supplier must approve the nomination within ten (10) Working Days unless the Supplier provides a reasonable explanation for rejecting the appointment. If the appointment is rejected then the Buyer may propose an alternative benchmarker. If the Parties cannot agree the appointment within twenty (20) days of the initial request for Benchmark review then a benchmarker shall be selected by the Chartered Institute of Financial Accountants.

3.1.7 The cost of a benchmarker shall be borne by the Buyer (provided that each Party shall bear its own internal costs of the Benchmark Review) except where the Benchmark Review demonstrates that the Benchmarked Service and/or the Benchmarked Deliverables are not Good Value, in which case the Parties shall share the cost of the benchmarker in such proportions as the Parties agree (acting reasonably). Invoices by the benchmarker shall be raised against the Supplier and the relevant portion shall be reimbursed by the Buyer.

## 3.2 Benchmarking Process

- 3.2.1 The benchmarker shall produce and send to the Buyer, for Approval, a draft plan for the Benchmark Review which must include:
  - (a) a proposed cost and timetable for the Benchmark Review;
  - (b) a description of the benchmarking methodology to be used which must demonstrate that the methodology to be used is capable of fulfilling the benchmarking purpose; and
  - a description of how the benchmarker will scope and identify the Comparison Group.
- 3.2.2 The benchmarker, acting reasonably, shall be entitled to use any model to determine the achievement of value for money and to carry out the benchmarking.
- 3.2.3 The Buyer must give notice in writing to the Supplier within ten (10) Working Days after receiving the draft plan, advising the benchmarker and the Supplier whether it Approves the draft plan, or, if it does not approve the draft plan, suggesting amendments to that plan (which must be reasonable). If amendments are suggested then the benchmarker must produce an amended draft plan and this Paragraph 3.2.3 shall apply to any amended draft plan.
- 3.2.4 Once both Parties have approved the draft plan then they will notify the benchmarker. No Party may unreasonably withhold or delay its Approval of the draft plan.
- 3.2.5 Once it has received the Approval of the draft plan, the benchmarker shall:
  - (a) finalise the Comparison Group and collect data relating to Comparable Rates. The selection of the Comparable Rates (both in terms of number and identity) shall be a matter for the Supplier's professional judgment using:
    - (i) market intelligence;
    - (ii) the benchmarker's own data and experience;
    - (iii) relevant published information; and
    - (iv) pursuant to Paragraph 3.2.6 below, information from other suppliers or purchasers on Comparable Rates;

- (b) by applying the adjustment factors listed in Paragraph 3.2.7 and from an analysis of the Comparable Rates, derive the Equivalent Data;
- using the Equivalent Data, calculate the Upper Quartile;
- (d) determine whether or not each Benchmarked Rate is, and/or the Benchmarked Rates as a whole are, Good Value.
- 3.2.6 The Supplier shall use all reasonable endeavours and act in good faith to supply information required by the benchmarker in order to undertake the benchmarking. The Supplier agrees to use its reasonable endeavours to obtain information from other suppliers or purchasers on Comparable Rates.
- 3.2.7 In carrying out the benchmarking analysis the benchmarker may have regard to the following matters when performing a comparative assessment of the Benchmarked Rates and the Comparable Rates in order to derive Equivalent Data:
  - the contractual terms and business environment under which the Comparable Rates are being provided (including the scale and geographical spread of the customers);
  - (b) exchange rates;
  - (c) any other factors reasonably identified by the Supplier, which, if not taken into consideration, could unfairly cause the Supplier's pricing to appear non-competitive.

### 3.3 Benchmarking Report

- 3.3.1 For the purposes of this Schedule "Benchmarking Report" shall mean the report produced by the benchmarker following the Benchmark Review and as further described in this Schedule;
- 3.3.2 The benchmarker shall prepare a Benchmarking Report and deliver it to the Buyer, at the time specified in the plan Approved pursuant to Paragraph 3.2.3, setting out its findings. Those findings shall be required to:
  - (a) include a finding as to whether or not a Benchmarked Service and/or whether the Benchmarked Deliverables as a whole are, Good Value;
  - (b) if any of the Benchmarked Deliverables are, individually or as a whole, not Good Value, specify the changes that would be required to make that Benchmarked Service or the Benchmarked Deliverables as a whole Good Value; and
  - (c) include sufficient detail and transparency so that the Party requesting the Benchmarking can interpret and understand how the Supplier has calculated whether or not the Benchmarked Deliverables are, individually or as a whole, Good Value.

Call-Off Schedule 16 (Benchmarking) Call-Off Ref:C52340 PD - NHS Pathways Crown Copyright 2018

> 3.3.3 The Parties agree that any changes required to this Contract identified in the Benchmarking Report shall be implemented at the direction of the Buyer in accordance with Clause 24 (Changing the contract).
Call-Off Schedule 18 (Background Checks) Call-Off Ref:C52340 PD - NHS Pathways Crown Copyright 2018

# Call-Off Schedule 18 (Background Checks)

## 1. When you should use this Schedule

This Schedule should be used where Supplier Staff must be vetted before working on the Contract.

## 2. Definitions

"Relevant Conviction" means any conviction listed in Annex 1 to this Schedule.

## 3. Relevant Convictions

- 3.1.1 The Supplier must ensure that no person who discloses that they have a Relevant Conviction, or a person who is found to have any Relevant Convictions (whether as a result of a police check or through the procedure of the Disclosure and Barring Service (DBS) or otherwise), is employed or engaged in any part of the provision of the Deliverables without Approval.
- 3.1.2 Notwithstanding Paragraph 3.1.1 for each member of Supplier Staff who, in providing the Deliverables, has, will have or is likely to have access to children, vulnerable persons or other members of the public to whom the Buyer owes a special duty of care, the Supplier must (and shall procure that the relevant Sub-Contractor must):
  - (a) carry out a check with the records held by the Department for Education (DfE);
  - (b) conduct thorough questioning regarding any Relevant Convictions; and
  - (c) ensure a police check is completed and such other checks as may be carried out through the Disclosure and Barring Service (DBS),

and the Supplier shall not (and shall ensure that any Sub-Contractor shall not) engage or continue to employ in the provision of the Deliverables any person who has a Relevant Conviction or an inappropriate record.

C52340 NHS Pathways 03 Call Off Schedules 1.0F 22Dec21

Call-Off Schedule 18 (Background Checks) Call-Off Ref:C52340 PD - NHS Pathways Crown Copyright 2018

# Annex 1 – Relevant Convictions

# Call-Off Schedule 20 (Call-Off Specification)

This Schedule sets out the characteristics of the Deliverables that the Supplier will be required to make to the Buyers under this Call-Off Contract

## Definitions

Expression or Acronym	Definition
BAU	Business as usual
UEC	Urgent and Emergency Care
GDS	Government Digital Service
SOW	Statement of Work
EAB Enterprise Architecture Board	
SNOMED Systemised Nomenclature of Medicine – Clinical Te	

## Call-Off Schedule 20 (Call-Off Specification)

## 1. Context

NHS Pathways Clinical Decision Support System is a suite of around 20 products used in Urgent and Emergency care settings. It is used over the telephone to support non-clinical call handlers in directing callers to the most appropriate level of care for the care needs of the patient at the time of the call. It does not seek to diagnose patients but assesses symptoms to direct them to the most appropriate care setting for further assessment where required. NHS Pathways content also 'powers' 111 online.

It is used by all NHS 111 telephone service providers in England, and approximately half of 999 Ambulance service providers. It currently supports the remote assessment of nearly 19 million triages per year. It is also used by clinicians to support their assessments.

Technical and development support for the NHS Pathways BAU service is provided via an 'in-house' service which is a blend of permanent staff and supplier resources.

The tooling that underpins NHS Pathways has been on a purely maintain-only basis for many years. Various areas for future improvement have been identified which were confirmed via a Discovery phase looking at the tooling and the associated IT systems and a single Alpha looking at future Data Models.

The reports and other outputs from the Discovery phase and the single Alpha provide a baseline for the work now required to re-platform, re-architect and re-tool NHS Pathways. The expectation is for a redevelopment of the primary components on NHS Pathways, potentially including:

- New data models
- New authoring (content) tooling
- New content-distribution approach (covering technical changes and therefore new business process)

## 2. Scope

This call-off contract is to support the delivery of the Buyers outcomes related to the Live Development and DevOps Delivery (including continuous improvement) of the NHS Pathways suite of products. This includes:

- Discovery work, proof of concepts and other technical validation and operational viability work as a precursor to larger change work being undertaken.
- Development and delivery of new features into live service. The Supplier will be responsible for delivering several changes, new features and functionality as prioritised on the product backlog.
- · Day-to-day operations including on-going maintenance.

This call-off contract will also include any activity associated with a handover of the contract from the incumbent suppliers to the new supplier (if supplier is different).

Re-platforming/re-architecture and re-tooling of NHS Pathways

This call-off contract is to support the delivery of the Buyers outcomes related to the re-platforming, re-architecture and re-tooling of NHS Pathways (known as Pathways re-platforming)

The objective of the call-off is to address the issues that have been confirmed via a Discovery phase looking at the tooling and the associated IT systems in 2020 and a single Alpha looking at future Data Models in early 2021. The details of this work carried out to date will be shared with the successful supplier as part of the initial replatforming SOW (see SOW 2).

These issues include-

- The pace at which new Pathways content can be safely authored (from a clinical perspective) and distributed via the current tools does not support agile ways of working.
- The data models and the associated IT systems/tools require modernisation to address the following-
  - They are inflexible, and complex to change.
  - They need 'future proofing' to enable NHS Pathways to take advantage of new and emerging technologies
- Development is required to bring the tooling up to the expectations of the organisations latest Software Engineering standards.

The services required will include-

- Proof of concepts and other technical validation and operational viability work (such as Alpha and Beta testing) building on previously completed Discoveries, as a precursor to the change work being undertaken.
- Development and delivery of the change work. The supplier will be responsible for delivering IT system redevelopment, potentially including new data models, new authoring tooling and a new content-distribution approach.

NHS Pathways have an overarching vision of what Pathways re-platforming needs to deliver in terms of a modernised system that addresses the identified issues and is 'fit for purpose' for the future, along with an associated high-level budget and expected timeframe.

However, the phases of work required to deliver that vision are flexible with the detailed scope and requirements for each phase only becoming clear on completion of the previous phase. This is typical of Agile working.

To that end, there will be several associated Statements of Work (SOWs) under the overall Call-off and within the scope of the budget allocated that the supplier will be required to deliver against to achieve each of the phases needed to meet the overarching vision.

.....

2.1. Geographical Scope

2.1.1. Target Geography

England

## 2.1.2. Delivery Geography

For BAU DevOps, the Supplier team are expected to attend 7 & 8 Wellington Place, Leeds. LS1 4AJ on most days, with the option to work remotely some of the time.

For re-platforming, the Supplier team are expected to work largely remotely, but with a secondary location of attending 7 & 8 Wellington Place, Leeds. LS1 4AJ to meet with the members of NHS Pathways and the Supplier BAU DevOps team.

2.1.3. Organisational Scope

NHS Pathways sits within the Urgent and Emergency Care (UEC) Directorate within NHS Digital. The UEC Directorate itself sits under the Product Directorate.

2.2. End Users

2.2.1. Target Sectors

arget Sectors details are contained in:	Annex 1
Addition of 'Urgent Emergency Care Sector' to the Health and Social Care Settings table in Annex 1.	

## 2.2.2. Target Users

Target users will be members of the Urgent and Emergency Care sub-Directorate within NHS Digital.

Re-platforming and retooling will focus on NHS Pathways staff (including clinicians) who are the primary users.

## 2.3. Stakeholders

Stakeholders will primarily be internal and members of the Urgent and Emergency Care sub-Directorate within NHS Digital. For a non-exhaustive list:

- UEC Directory of Services
- 111 Online

In addition, for Pathways re-platforming, the following stakeholders outside of NHS Digital will be involved:

- Interested parties in NHSx and NHS England Urgent and Emergency Care
- · The host system suppliers who embed the Pathways content
- Indirectly the 111 & 999 providers who use the Pathways content

## 2.4. Timing Scope

This call-off contract is dependent on funding being agreed for the period of its duration, which in turn may be impacted by government, departmental and organisational priorities. Funding is requested on a yearly basis.

For Pathways re-platforming, it is envisaged that the data model & the associated IT systems/tools that are built will future-proof NHS Pathways for at least 5 years, subject to ongoing continuous improvements under the BAU service.

2.5. Life-Cycle Scope

For the BAU service, this will cover running and maintaining but no need of handover or exit transition until the end of the call-off period.

For Pathways re-platforming, this will encompass the full set of GDS agile build phases including some further Discovery, but mainly Alpha, Beta and migration to the Live service. With the same supplier undertaking both support for the BAU service and the re-platforming work, this will enable the effective transition of the replatformed service into Live operation.

The SOW's for each phase of the re-platforming will identify the outputs that the Supplier is required to produce at the end of that phase to ensure a smooth transition from phase to phase with minimal delay.

- 2.6. Technical Scope
  - 2.6.1. Core Technical Capabilities

Technology Capability details are contained in:	Annex 2
NHS Pathways have built several bespoke applications to support the programme and its stakeholders.	
All websites, web services and windows applications are C#.Net and MS-SQL.	
The DevOps team will focus on these from a database and application perspective.	
The re-platforming team will be looking much more broadly, covering several areas in Annex 2, such as architecture and data modelling as well as the other areas.	



## 2.6.2. Diagram of the Technical Scope

Note that this diagram is provided to help Suppliers position Pathways within the wider UEC system and is absolutely not the Technical Scope of the Pathways replatforming itself as that will only become clear following the initial phases. The Pathways scope is covered in the diagram by the 111 and Emergency (999) Systems, Intelligent Data Tool and to a lesser degree the 111 Online system that uses NHS Pathways.

2.6.3. Technical Scope Exclusions

### None

## 2.6.4. Interoperability Considerations

There will be some focus on how interoperability between the 111 and 999 systems and the Urgent and Emergency Care Directory of Services interact.

It is expected, but not certain, that a new API based content-distribution approach will be an outcome of re-platforming. Building and implementing such a new API would require interoperability work. The NHS Digital Urgent & Emergency Care Interoperability team would provide the oversight for any interoperability work required.

## 2.6.5. Technical Constraints

For the DevOps team, NHS Pathways have built several websites, web services and windows applications. All run on C# .Net and SQL. Currently hosted on on-prem virtual instances, but work is beginning on migration to AWS.

However, there are no constraints for the re-platforming work as all technical solutions are for consideration.

### 2.7. Requirements

For the DevOps requirements:

- Iterate on a sprint basis through existing user stories on the C# .Net/SQL products that NHS Pathways built and support (including websites, webservices, windows applications and databases).
- Release changes to the various NHS Pathways websites, webservices, windows applications and databases.
- Diagnose and raise issues and to deploy new versions of a 3rd party application known as the NHS Pathways Intelligent Data Tool (IDT).

For Pathways re-platforming, the Buyer has a requirement to secure a Supplier to:

- Deliver development and associated services to the Buyer pursuant to Statement
  of Works (SOWs) agreed under the Call-Off Contract.
- Deliver such services across a range of NHS Pathways data models, systems and tools.
- Potentially deliver such services across different locations (as detailed in the applicable SOWs); and
- Deliver such services in a manner compliant with the terms and conditions of the Call-Off Contract, including without limitation the Buyer's requirements as further detailed in the contract Schedules.

The Buyer expects the Supplier to work with the Buyer team as follows:

- Work closely with NHS Digital in-house teams and contractors including solutions assurance teams and governance bodies such as the Enterprise Architecture Board (EAB), Technical review board and Platforms directorate.
- The Buyer will have a client-side engagement team in place consisting of roles such as: Delivery & Contract Manager, Head of Development, Senior Product Roles, Technical Programme Managers, Technical Architects, Project & Delivery Managers, Developers, Data Architects and Business Analysts.
- There are several existing NHS Digital staff across the Pathways programme who may from time to time need to be incorporated into the delivery teams.

The requirements appropriate to this call-off contract are to work proactively with the Buyer to provide an efficient End to End Services capability undertaking the full set of GDS agile phases of some further Discovery, but predominantly Alpha, Beta and delivering into Live service to enable the required transformation of the Pathways data model, tooling & associated IT systems.

The ongoing running of the Live service once bedded in is not in scope for Pathways re-platforming as this will be the responsibility of the BAU team.

The Supplier shall be required to adopt an agile iterative development methodology and work with the Buyer (and potentially other Buyer development suppliers).

2.8. Business Outcomes

	Annex 3
There are a number of proposed themes for Pathways re-platforming which indicate the expected business outcomes. Due to the high-level	
natures of these themes, which will evolve and become clearer as the	
initial phases of re-platforming are undertaken, the use of the Epics	
template in Annex 3 is not appropriate at this juncture.	
<ul> <li>Simplified authoring (clinical content)</li> </ul>	
<ul> <li>Simplified authoring (clinical content)</li> <li>Simplified reporting (data)</li> </ul>	
<ul> <li>SNOMED coded (standardised) inputs and outputs</li> </ul>	
<ul> <li>Simplified distribution of content (releases)</li> </ul>	
Easier updating of content	
<ul> <li>Simplified outcome mapping to services available (linking to</li> </ul>	
Directory of Services)	
These expected business outcomes give a high-level overview of the	4
current vision for NHS Pathways on completion of the re-platforming	
work and delivering these outcomes is designed to address the issues	
highlighted in Section 2, Scope.	
The expected outcomes of Simplified authoring, Simplified reporting,	
Easier updating of content and Simplified distribution of content	
(releases) will benefit the NHS Pathways internal stakeholders	4
directly, notably the clinical authors and release team, and indirectly	
the provider & supplier end-consumers through our ability to deliver	
improvements to the Pathways content to them more quickly.	
The expected outcomes of SNOMED coded (standardised) inputs	
and outputs and Simplified outcome mapping to services available will	
provide strategic and interoperability benefits to the wider urgent and	
emergency care system.	

2.9. Technical Exclusions

N/A

## 3. Deliverables

nitial Product Backlog Item List overview is contained in:	Annex 4
--	---------

3.1. Functional and Non-Functional Requirements

For Pathways re-platforming, the information gathered to date on functional and nonfunctional requirements under the Discovery and Alpha initiatives, referenced in section 2, will be shared with the successful supplier as part of the initial replatforming SOW (see SOW 2).

3.2. Target Operating Model

### NHS Pathways in Urgent and Emergency Care

As stated in the Context section above, the NHS Pathways Clinical Decision Support System is a suite of around 20 products used in Urgent and Emergency care settings. An overview of the current Urgent and Emergency care environment and operating model can found on the NHS England web-site https://www.england.nhs.uk/urgent-emergency-care/.

Pathways is used over the telephone to support non-clinical call handlers in directing callers to the most appropriate level of care for the care needs of the patient at the time of the call. It does not seek to diagnose patients but assesses symptoms to direct them to the most appropriate care setting for further assessment where required.

It is used by all NHS 111 telephone service providers in England, and approximately half of 999 Ambulance service providers. It currently supports the remote assessment of nearly 19 million triages per year. NHS Pathways is used by clinicians to support their assessments and the Pathways content also 'powers' 111 online and is used to support ED streaming and re-direction.

When used by NHS 111 & 999 services under the current operating model, the NHS Pathways clinical content is embedded by host system suppliers in their systems which provide the front-line services such as demographics data capture, computer aided despatch and service directory searching. An overview of the complex, interconnected web of technologies that support Urgent and Emergency care is described diagrammatically in section 2.6.2.

The NHS Urgent and Emergency care (UEC) environment and operating model that Pathways operates in is dynamic and fast-changing. When Pathways began in the early 2000's, UEC looked very different to the current model with recent changes that Pathways has had to be adapted to including the move from 111 to Integrated Urgent Care (IUC) with Clinical Assessment Services (CAS) and the introduction of 111 First to encourage patients to call 111 for triage before attending accident and emergency (A&E) services.

Further significant changes to the operating model are inevitable as the NHS grapples with the challenge of the increasing demand on UEC services and the environment that NHS Pathways operates in will look very different over the next 3 to 5 years.

By delivering, through the re-platforming element of this call off, the 6 business outcomes listed in section 2.8 and addressing the issues outlined in section 2.0, the NHS Pathways Clinical Decision Support System will be much better able to respond to the rapid pace of change in UEC and will remain a core component of the UEC target operating model for the foreseeable future.

## Potential NHS Pathways Target Operating Model following Re-platforming from Discovery Work -

With regards to the potential Target Operating Model for the NHS Pathways Clinical Decision Support System itself on completion of the re-platforming work, whilst we would expect the successful supplier to design and propose this, the Discovery activity carried out to date included a proposed target operating model which is pasted in below to provide an indication of the direction of travel:-

C52340 NHS Pathways 03 Call Off Schedules 1.0F 22Dec21

Call-Off Schedule 20 (Call-Off Specification) Call-Off Ref: Crown Copyright 2018

		100	KPI P	Organization	Infe	
and the second se	reation and delive		NI P	Organisation	2//	
Deliverit	ng required conten	nt	Comercian and	Business C	Business	0006
Clinical a	issurance and aut	omated testing	There are provided in the former	Steering proup divice authors		ter/consumer
Content	research and pres	paration	Plantame	THE PARTY A	Occument	
Workflow	W (president)	And the Association of Concerning of	and the second	the second second	Delivery Al	PE yourcos
technology		0-0	Constituter Salars	Name International	Technology	(E)
	ted authoring sup	CO	Restaurantial Designation	States in case of the local division of the	Cloud money	CONTRACTOR CONTRACTOR
	tic unit and regres	and the state of t				strate travers creation)
	The second s	audit & reporting	2.1	Aven N		SOOIS (traing, release notes)
and the second se	delivery (portable				Delivery me	charism .
0.00000000	Delige and the second	Contraction of the second s				STREET, STREET
conditioner	H Arsourch and	Create and Assure comment			conunt > D	Alvery pipeline
	6			Atume		
	Capability			1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Marcal	
		10000		- / / / /	Manage	ment systems
lusiness		8987	Contra Contra	advance.		10
Assurance	<ul> <li>quality content</li> </ul>	6 MA	Contract of the local division of the local	THE REAL PROPERTY IN CONTRACTOR OF THE PROPERTY INTERPOPERTY INTER	Business	-629-
Iterative c	reation, peer revi	ew and assurance.	1 12		Clinical go	vernance
	A TOTAL CONTRACTOR OF A TOTAL CONTRACTOR OF A TOTAL CONTRACTOR OF A TOTAL CONTRACTOR OF A TOTAL CONTRACTOR A TO	e to create content	67	2	Release to	
Scheduled	and Continuous	release of content		Training 8-8	Steering a	POUD UNA
		6	Cloud	Embed content	( 000.00	eat when
technology	1000000 C	E I	Authoring support	Implementation testing	at 1977	50000
Pipeline A	A DECEMBER 1	Integr	ated application suite	Consumer content as a	service	15 31
Applicatio			Infrastructure	Supplier/partn	ers	CON 1
Test-drive	e development					
~ T	Iterative Au Iterative cre Real time in	consumer ava thoring proce ation and assu teractive creat	iss irance at the atom ion and dynamic v	visualisation of con	sumerview	
H ®	Web based Iterative Au Iterative cre Real time in Ability to en Real time fe Content de Continuous Content put	consumer ava thoring proce ation and assu- teractive creat hance content redback and as <b>livery</b> pipeline based olished via stan	rance at the atom ion and dynamic v logic and custom isisted content cre delivery of conte idard API, contain	ic level risualisation of con ise for specific use	sumerview case najor releases pecific use case	
© H @ A	Web based Iterative Au Iterative cre Real time in Ability to en Real time fe Content de Content de Content put Portable ap Testing	consumer ava thoring proce- ation and assu- teractive creat hance content redback and as fivery pipeline based bished via stan plications for tr	rance at the atom ion and dynamic v logic and custom isisted content cre delivery of conte idard API, contain	ic level risualisation of con ise for specific use sation nt plus scheduled r erised to support s ucing of content (st	sumerview case najor releases pecific use case	
ی ۳ ۳	Web based Iterative Au Iterative cre Real time in Ability to en Real time fe Content de Continuous Content put Portable ap Testing Continuous	consumer ava thoring proce- ation and assu- teractive creat hance content redback and as fivery pipeline based bished via stan plications for tr	rance at the atom ion and dynamic v logic and custom isisted content cre delivery of conte dard API, contain iage/PaCCS surfa	ic level risualisation of con ise for specific use sation nt plus scheduled r erised to support s ucing of content (st	sumerview case najor releases pecific use case	
<ul> <li>(a)</li> <li>(b)</li> <li>(c)</li> <li>(c)</li></ul>	Web based Iterative Au Iterative cre Real time in Ability to en Real time fe Content de Continuous Content put Portable ap Testing Continuous Automated Automated	consumer ava thoring proce- ation and assu- teractive creat hance content redback and as <b>livery</b> pipeline based bished via stan plications for tr assurance of or maintenance validation and	rss irance at the atom ion and dynamic v logic and custom isisted content cre idelivery of conte idard API, contain iage/PaCCS surfa content during cre testing	ic level risualisation of con ise for specific use sation nt plus scheduled r erised to support s iccing of content (st ation	sumerview case najor releases pecific use case yle, business log	
<ul> <li>(b)</li> <li>(c)</li> <li>(c)</li></ul>	Web based Iterative Au Iterative cre Real time in Ability to en Real time fe Content de Continuous Content put Portable ap Testing Continuous Automated Automated	consumer ava thoring proce- ation and assu- teractive creat hance content redback and as <b>livery</b> pipeline based bished via stan plications for tr assurance of or maintenance validation and	rss irance at the atom ion and dynamic v logic and custom isisted content cre idelivery of conte idard API, contain iage/PaCCS surfa content during cre testing	ic level risualisation of con ise for specific use sation nt plus scheduled r erised to support s ucing of content (st	sumerview case najor releases pecific use case yle, business log	
	Web based Iterative cre Real time in Ability to en Real time fe Content de Continuous Content put Portable ap Testing Continuous Automated Automated Minimal imp	consumer ava thoring proce ation and assu- teractive creat hance content redback and as <b>livery</b> pipeline based olished via stan plications for tr assurance of or maintenance validation and sementation te	rance at the atom ion and dynamic v logic and custom isisted content cre idard API, contain iage/PaCCS surfa content during cre testing sting (published a	ic level risualisation of con ise for specific use sation nt plus scheduled r erised to support s iccing of content (st ation	sumerview case najor releases pecific use case yle, business log	
	Web based Iterative cre Real time in Ability to en Real time fe Content de Continuous Content put Portable ap Testing Continuous Automated Automated Minimal imp	consumer ava thoring proce ation and assu- teractive creat hance content redback and as <b>livery</b> pipeline based olished via stan plications for tr assurance of or maintenance validation and is dementation ter data structure	rance at the atom ion and dynamic v logic and custom isisted content cre idard API, contain iage/PaCCS surfa content during cre testing sting (published a	ic level risualisation of con ise for specific use sation nt plus scheduled r erised to support s iccing of content (st ation	sumerview case najor releases pecific use case yle, business log	
	Web based Iterative Au Iterative cre Real time in Ability to en Real time fe Content de Continuous Content put Portable ap Testing Continuous Automated Automated Minimal imp Enhanced Import/expo	consumer ava thoring proce ation and assu- teractive creat hance content redback and as fivery pipeline based olished via stan plications for tr assurance of o maintenance validation and is dementation ter data structure rt assertion an	rance at the atom ion and dynamic v logic and custom isisted content cre idard API, contain iage/PaCCS surfa content during cre testing sting (published a	ic level risualisation of con ise for specific use sation nt plus scheduled r erised to support s iccing of content (st ation	sumerview case najor releases pecific use case yle, business log	
	Web based Iterative Au Iterative cre Real time in Ability to en Real time fe Content de Continuous Content put Portable ap Testing Continuous Automated Automated Minimal imp Enhanced Import/expo	consumer ava thoring proce ation and assu- teractive creat hance content redback and as <b>livery</b> pipeline based olished via stan plications for tr assurance of or maintenance validation and to lementation ter data structure rt assertion an CT coding	rance at the atom ion and dynamic v logic and custom isisted content cre idard API, contain iage/PaCCS surfa content during cre testing sting (published a d demographics	iic level risualisation of con ise for specific use sation nt plus scheduled r erised to support s icing of content (st ation nd consumed via A	sumerview case najor releases pecific use case yle, business log	
	Web based Iterative Au Iterative cre Real time in Ability to en Real time fe Content de Content de Continuous Content put Portable ap Testing Continuous Automated Automated Minimal imp Enhanced Import/expo SNOMED C Enrich conti	consumer ava thoring proce ation and assu- teractive creat hance content redback and as <b>livery</b> pipeline based olished via stan plications for tr assurance of or maintenance validation and is lementation ter data structure of assertion an CT coding ent for specific	rance at the atom ion and dynamic v logic and custom isisted content cre idard API, contain iage/PaCCS surfa content during cre testing sting (published a d demographics use case (CDSS,	ic level risualisation of con ise for specific use sation nt plus scheduled r erised to support s iccing of content (st ation	sumerview case najor releases pecific use case yle, business log	
	Web based Iterative Au Iterative cre Real time in Ability to en Real time fe Content de Content de Continuous Content put Portable ap Testing Continuous Automated Automated Minimal imp Enhanced Import/expo SNOMED C Enrich conti	consumer ava thoring proce ation and assu- teractive creat hance content redback and as <b>livery</b> pipeline based olished via stan plications for tr assurance of or maintenance validation and is lementation ter data structure of assertion an CT coding ent for specific	rance at the atom ion and dynamic v logic and custom isisted content cre idard API, contain iage/PaCCS surfa content during cre testing sting (published a d demographics	iic level risualisation of con ise for specific use sation nt plus scheduled r erised to support s icing of content (st ation nd consumed via A	sumerview case najor releases pecific use case yle, business log	
	Web based Iterative Au Iterative Cre Real time in Ability to en Real time fe Content de Continuous Content put Portable ap Testing Continuous Automated Automated Minimal imp Enhanced Import/expo SNOMED C Enrich conti	consumer ava thoring proce- ation and assu- teractive creat hance content redback and as <b>livery</b> pipeline based bished via stan plications for tr assurance of or maintenance validation and to be mentation te data structure rt assertion an CT coding ent for specific creation and re	rance at the atom ion and dynamic v logic and custom isisted content cre idard API, contain iage/PaCCS surfa content during cre testing sting (published a d demographics use case (CDSS,	iic level risualisation of con ise for specific use sation nt plus scheduled r erised to support s icing of content (st ation nd consumed via A	sumerview case najor releases pecific use case yle, business log	
	Web based Iterative cre Real time in Ability to en Real time fe Content de Continuous Content put Portable ap Testing Continuous Automated Automated Minimal imp Enhanced Import/expo SNOMED C Enrich cont Support for Document	consumer ava thoring procession ation and assu- teractive creat hance content redback and as <b>livery</b> pipeline based bished via stan plications for tr assurance of or maintenance validation and to be mentation te data structure of assertion and CT coding ent for specific creation and re management	rance at the atom ion and dynamic v logic and custom isisted content cre idelivery of conte- idard API, contain iage/PaCCS surfa content during cre testing sting (published a d demographics use case (CDSS, ilease processes	iic level risualisation of con ise for specific use sation nt plus scheduled r erised to support s icing of content (st ation nd consumed via A	sumerview case najor releases pecific use case yle, business log	
	Web based Iterative Au Iterative Cre Real time in Ability to en Real time fe Content de Continuous Content put Portable ap Testing Continuous Automated Automated Minimal imp Enhanced Import/expo SNOMED C Enrich cont Support for Document Core clinica	consumer ava thoring procession ation and assu- teractive creat hance content redback and as <b>livery</b> pipeline based bished via stan plications for tr assurance of or maintenance validation and to be mentation te data structure of assertion an CT coding ent for specific creation and re management il reference mo	rss irance at the atom ion and dynamic v logic and custom isisted content cre idelivery of conte- idard API, contain iage/PaCCS surfa content during cre testing sting (published a d demographics use case (CDSS, ilease processes del	iic level risualisation of con ise for specific use sation nt plus scheduled r erised to support s icing of content (st ation nd consumed via A	sumerview case najor releases pecific use case yle, business log	
	Web based Iterative Au Iterative Cre Real time in Ability to en Real time fe Content de Continuous Content put Portable ap Testing Continuous Automated Automated Minimal imp Enhanced Import/expo SNOMED C Enrich cont Support for Document Core clinica	consumer ava thoring procession ation and assu- teractive creat hance content redback and as <b>livery</b> pipeline based bished via stan plications for tr assurance of or maintenance validation and to be mentation te data structure of assertion and CT coding ent for specific creation and re management	rss irance at the atom ion and dynamic v logic and custom isisted content cre idelivery of conte- idard API, contain iage/PaCCS surfa content during cre testing sting (published a d demographics use case (CDSS, ilease processes del	iic level risualisation of con ise for specific use sation nt plus scheduled r erised to support s icing of content (st ation nd consumed via A	sumerview case najor releases pecific use case yle, business log	
	Web based Iterative cre Real time in Ability to en Real time fe Content de Continuous Content put Portable ap Testing Continuous Automated Automated Minimal imp Enhanced Import/expo SNOMED C Enrich conti Support for Document Core clinica Integrated a	consumer available consumer available content setuctive creat hance content setucated as the content setucated as the content setucated as the content setucated as the content setucations for the content setucation and the content setuca	rss irance at the atom ion and dynamic v logic and custom isisted content cre idelivery of conte- idard API, contain iage/PaCCS surfa content during cre testing sting (published a d demographics use case (CDSS, ilease processes del	iic level risualisation of con ise for specific use sation nt plus scheduled r erised to support s icing of content (st ation nd consumed via A	sumerview case najor releases pecific use case yle, business log	
	Web based Iterative cre Real time in Ability to en Real time fe Content de Continuous Content put Portable ap Testing Continuous Automated Automated Minimal imp Enhanced Import/expo SNOMED C Enrich conto Support for Document Core clinica Integrated a	consumer available consumer available content sedback and assurance of content sedback and as the content sedback and a structure of the content sedback and the content sedback and the content sedback and the content sedback as the content sedback as the content sedback as the content sedback as the content sedback and the content sedback and the content sedback as the co	rance at the atom ion and dynamic v logic and custom isisted content cre dard API, contain iage/PaCCS surfa content during cre testing sting (published a d demographics use case (CDSS, lease processes del d repository	iic level isualisation of con ise for specific use sation int plus scheduled r erised to support s icing of content (st) ation ind consumed via A PaCCS, 111 online	sumer view case najor releases pecific use case rie, business log (PI) e etc)	pic, and content)
- State (1997)	Web based Iterative cre Real time in Ability to en Real time fe Content de Continuous Content put Portable ap Testing Continuous Automated Automated Minimal imp Enhanced Import/expo SNOMED C Enrich conto Support for Document Core clinica Integrated a	consumer available consumer available content setuctive creat hance content setuback and as tivery pipeline based olished via stam plications for transurance of consumeration terestion and setuctions for transitional constructions for transitional construction and setuction and set	rance at the atom ion and dynamic v logic and custom isisted content cre dard API, contain iage/PaCCS surfa content during cre testing sting (published a d demographics use case (CDSS, lease processes del d repository	iic level isualisation of con setion int plus scheduled r erised to support s icing of content (st) ation ind consumed via A PaCCS, 111 online	sumer view case najor releases pecific use case rie, business log (PI) e etc)	pic, and content)
	Web based Iterative cre Real time in Ability to en Real time fe Content de Continuous Content put Portable ap Testing Continuous Automated Automated Minimal imp Enhanced Import/expo SNOMED C Enrich conto Support for Document Core clinica Integrated a	consumer available consumer available content setuctive creat hance content setuback and as tivery pipeline based olished via stam plications for transurance of consumeration terestion and setuctions for transitional constructions for transitional construction and setuction and set	rance at the atom ion and dynamic v logic and custom isisted content cre dard API, contain iage/PaCCS surfa content during cre testing sting (published a d demographics use case (CDSS, lease processes del d repository	iic level isualisation of con setion int plus scheduled r erised to support s icing of content (st) ation ind consumed via A PaCCS, 111 online	sumer view case najor releases pecific use case rie, business log (PI) e etc)	pic, and content)
- State (1997)	Web based Iterative cre Real time in Ability to en Real time fe Content de Content put Portable ap Testing Continuous Automated Automated Minimal imp Enhanced Import/expo SNOMED C Enrich cont Support for Document Core clinica Integrated a Morkflow n	consumer ava thoring procession ation and assu- teractive creat hance content redback and as <b>livery</b> pipeline based olished via stan plications for tr assurance of of maintenance validation and re- validation and re- data structure of assertion and CT coding ent for specific creation and re- management inference mo- suthoring recom- management realize release pri- hange candida	rance at the atom ion and dynamic v logic and custom isisted content cre dard API, contain iage/PaCCS surfa content during cre testing sting (published a d demographics use case (CDSS, lease processes del d repository	iic level isualisation of con setion int plus scheduled r erised to support s icing of content (st) ation ind consumed via A PaCCS, 111 online	sumer view case najor releases pecific use case rie, business log (PI) e etc)	pic, and content)
1002/001X	Web based Iterative Au Iterative cre Real time in Ability to en Real time fe Content de Content de Content put Portable ap Testing Continuous Automated Automated Minimal imp Enhanced Import/expo SNOMED C Enrich cont Support for Document Core clinica Integrated a Integrated a	consumer ava thoring procession ation and assu- teractive creat hance content redback and as <b>livery</b> pipeline based olished via stan plications for tr assurance of of maintenance validation and re- validation and re- data structure of assertion and CT coding ent for specific creation and re- management inference mo- suthoring recom- management realize release pri- hange candida	rance at the atom ion and dynamic v logic and custom isisted content cre dard API, contain iage/PaCCS surfa content during cre testing sting (published a d demographics use case (CDSS, lease processes del d repository	iic level isualisation of con setion int plus scheduled r erised to support s icing of content (st) ation ind consumed via A PaCCS, 111 online	sumer view case najor releases pecific use case rie, business log (PI) e etc)	pic, and content)
2022/03PX	Web based Iterative Au Iterative Cre Real time in Ability to en Real time fe Content de Continuous Content put Portable ap Testing Continuous Automated Automated Automated Minimal imp Enhanced Import/expo SNOMED C Enrich conti Support for Document Core clinica Integrated a Morkflow n Integrated a	consumer ava thoring procession ation and assu- teractive creat hance content redback and as <b>livery</b> pipeline based bished via stan plications for tr assurance of or maintenance validation and re- validation and re- data structure of assertion and re- management il reference mo- suthoring recom- management assertion and re- management lireference mo- suthoring recom- management assertion and re- management assertion assertion and re- management assertion and re- management assertion	rance at the atom ion and dynamic v logic and custom isisted content cre idard API, contain iage/PaCCS surfa content during cre testing sting (published a d demographics use case (CDSS, ilease processes del d repository ocess with author ite selection and p	iic level isualisation of con setion int plus scheduled r erised to support s icing of content (st) ation ind consumed via A PaCCS, 111 online	sumer view case najor releases pecific use case rie, business log (PI) e etc)	pic, and content)
- State (1997)	Web based Iterative Au Iterative Cre Real time in Ability to en Real time fe Content de Continuous Content put Portable ap Testing Continuous Automated Automated Automated Minimal imp Enhanced Import/expo SNOMED C Enrich conti Support for Document Core clinica Integrated a Morkflow n Integrated a	consumer ava thoring procession ation and assu- teractive creat hance content redback and as <b>livery</b> pipeline based olished via stan plications for tr assurance of of maintenance validation and re- validation and re- data structure of assertion and CT coding ent for specific creation and re- management inference mo- suthoring recom- management realize release pri- hange candida	rance at the atom ion and dynamic v logic and custom isisted content cre idard API, contain iage/PaCCS surfa content during cre testing sting (published a d demographics use case (CDSS, ilease processes del d repository ocess with author ite selection and p	iic level isualisation of con setion int plus scheduled r erised to support s icing of content (st) ation ind consumed via A PaCCS, 111 online	sumer view case najor releases pecific use case rie, business log (PI) e etc)	pic, and content)

**Content Delivery** 

#### **Target Operation Model**

### **Content Creation**

- · Iterative inclusive content creation and assurance
- Integrated application suite working from a common data layer
- Integrated document management
- Integrated workflow and backlog management
- Remove external testing and training dependency from critical path

#### **Current Operation Model**

#### **Content** Creation

- Pragmented process and tools
- Fragmented and strachared support process and document management
- CSV equal
   Fortable App (SOLOPWW) for Consult/Hecka

Content Delivery

COV Portalia App (SOLO) with content for tasts, UAS and maning

· CI/CD pipeline for CSV published via API

validation/training

Enhanced data via API for specialist use cases.

Gold Standard reference application for external

#### **Testing and Assurance**

сом	том
<ul> <li>Fragmented</li> <li>Long delay between authoring and checking</li> <li>Large number of changes to check and fix</li> <li>Content too complex to check (not everything tested)</li> <li>Testing repeated by providers and suppliers</li> <li>Manual testing, multiple tools and methods</li> <li>High test failure rate</li> </ul>	<ul> <li>High level of inherent confidence in the published content</li> <li>Iterative creation and assurance, peer review and sign off</li> <li>Automatic integrity check of the data layer</li> <li>Rules-based authoring assistance to validate creation logic</li> <li>Intelligent assurance models to maintain the content</li> <li>Completely remove external testing from the critical path</li> <li>Basic (semi-automated) implementation testing</li> <li>Gold standard reference application and content (Consumer)</li> </ul>

CSV builds and clinical checks are late in the process and can lead to a lot of rework and subsequent testing. User acceptance test scripts are manually created and focus on known changes rather than unplanned changes, resulting in some avoidable issues being experienced in live use. Continuous testing and clinical assurance and automated support throughout the authoring process will improve output quality and end-user confidence in the published content. Semi-automation will support basic implementation regression testing, and significantly reduce the impact on external processes may result from increased implementation frequency. The Consumer will enable remote access from the provider site acting as the gold standard against which to cross-check implantation test results.

#### Tools

сом	том
Disparate and Fragmented     Mixed technologies     Monolithic .NET application     Complex to learn and master     Barrier to increase release frequency     Barrier to configuration management and unit     testing     Isolated development environment ino promotion to     hell     Low resistance to errors     Inherent waterfall approach     Manual     Bespoke and cross-functional	<ul> <li>Native Cloud technology</li> <li>Standard reusable APis and Data Repository</li> <li>An integrated suite of applications working from a shared data layer</li> <li>Support for iterative content creation, assurance and release</li> <li>Realtime testing and feedback to support authoring process</li> <li>Ability to dynamically edit content in text or graphical mode and display user view</li> <li>Ability to enhance content via dedicated applications</li> <li>Support for 'linear pathway' creation</li> <li>Future proof by design</li> <li>Easy to use</li> </ul>

Framework Ref: RM6221 Project Version: v1.0 Model Version: v3.0

#### 177 (22/12/2021)

An integrated set of applications (and underlying data structure) will provide the most significant opportunity for improvement of pathways content, enabling real time testing and clinical assurance of content as it is created.

#### **Content creation**

сом	том
<ul> <li>Multiple sources of candidates</li> <li>Fragmented tools and processes impede content creation</li> <li>Manual configuration management</li> <li>Too many changes per release</li> <li>Complex data set; difficult to maintain and improve</li> <li>Build and testing too late in the process</li> <li>Complex pyramid approach to development</li> <li>Independent processes for managing reference material</li> </ul>	<ul> <li>interactive tools on a common data layer</li> <li>iterative authoring and assurance of an immutable change</li> <li>intelligent content to support advanced use cases</li> <li>integrated workflow and document management</li> <li>Closely coupled with content delivery</li> <li>Fast and responsive</li> <li>Oriented to MDT and peer review and sign off</li> <li>Standard taxonomy (SNOMED)</li> </ul>

Current tools and processes impede content creation leading to duplication of effort, content, errors, and rework. Content creation is planned to use agile principles, but the tools and process constrain it to a waterfall process. Moving to an integrated application suite will support a faster iterative content creation and assurance approach. Changes to the data layer will enable the development of more intelligent content and access to a broader market.

#### **Content delivery**

сом		том
	Batched release in flat files External testing Batch training before implementation Suppliers need to import the content, build customer interface to the content, and test it for each release. Must update SOLO the for UAT testing. Manually created UAT, release scripts and tech specs. Frequent DoS update	<ul> <li>Continuous pipeline on new content published via standard API (existing)</li> <li>Tailored to specific use case</li> <li>Supplementary data to support advanced use cases</li> <li>Portable applications for specific use cases</li> <li>Portable triage application as a service</li> <li>Standard-based input/output (Assertions, outcome, acuity)</li> </ul>

Content will be published in a continuous pipeline as it's approved via a standard API. This can be implemented immediately or batched into scheduled releases. The content can also be delivered embedded in a portable application. Different versions of content and application can be published via the same API to be consumed based on rules such as training requirement, skill level etc. Additional supporting content can be published via a standard API in a way to support many specialist use-cases such as PaCCS, other CDSS, 111 online etc. To support multiple rules-based entry-points, the content will be coded, flexible, configurable. It will use static referential data to reduce the impact of other systems such as DoS.

A portable triage application with the latest content can be configured to be consumed as a service call by a host application, enabling NHS pathway to control the UX/UI as well as the content.

#### **Authoring Support**

OM	TOM
<ul> <li>Inconsistent management of references and sources</li> <li>Multiple repositories for reports, progress notes, audit</li> <li>Inconvenient planning and dev environment</li> <li>Multiple sources of backlog candidates (issues)</li> <li>Manual duplication of data from external systems</li> <li>No core clinical reference data set</li> <li>Disparate store of clinical documentation</li> <li>Multiple workflow processes</li> </ul>	<ul> <li>Integration with release process and tools (backlog, workflow, reporting, dashboards)</li> <li>Document registry and repository (Inc. document workflow)</li> <li>Core clinical reference</li> <li>Advanced authoring techniques</li> <li>Segment reuse</li> <li>Automated changes</li> <li>Bi and dashboard feedback</li> </ul>

There will be a seamless integration between release and authoring workflow with full report and dashboards. Bespoke forms will capture the necessary information for each change candidate, to support its prioritisation, and grouping into themes. A seamless process will manage all clinical references and documentation created by the authoring team to prepare clinical content, recording the evidence and reasoning behind the investigation and resultant outcomes and assertions. A portable triage application with the latest content can be configured to be consumed as a service call by a host application, enabling NHS pathway to control the UX/UI as well as the content.

COM	TOM
<ul> <li>Compound codes</li> <li>Inflexible</li> <li>Duplicate content</li> <li>Complex</li> <li>Limited reuse of content</li> <li>Does not support custom views of content</li> <li>Tied to CSV export</li> </ul>	<ul> <li>Immutable content to support configuration management</li> <li>Support release management</li> <li>Static referential data</li> <li>Support for standard API, codes, and other content consumer models</li> <li>Support an integrated toolset.</li> <li>Coded content and GUID</li> <li>Automated assurance and maintenance</li> <li>Intelligent content - Support import/export of 'assertions.'</li> </ul>

The data structure will be changed to support the new tooling and processes and to provide a richer view of the clinical content presented in a standard way with standard taxonomy. This is critical to provide content as a service to multiple CDSS.

#### **Release Management**

сом	том
A multi-step, manual process     Fragmented backlog management and reporting     Time boxed, waterfall (lockdown step)     Separate clinical and release workflow management     External processes on clinical path     Batched release, manual training, testing and release notes generation	<ul> <li>Integrated backlog management and workflow</li> <li>CI/CD pipeline based on content and requirements</li> <li>Automated release documentation</li> </ul>

A consolidated set of business processes will manage a change candidate through to delivery to the end-user, with full visibility and transparency

#### **Technology Layer**

DM (Software and infrastructure)	том
<ul> <li>Currently, a mixture of tools and processes based on different technologies and infrastructure.</li> <li>Older software versions (MindFusion, SQL server)</li> <li>Hosted/Online and desktop-based</li> <li>Mixed development tools and languages</li> <li>Low level of integration, proprietary tools.</li> <li>Incompatible with NHSD architectural principles</li> <li>Highly coupled architecture</li> </ul>	<ul> <li>Native Azure cloud technology</li> <li>Reusable components and De-coupled architecture</li> <li>TDD software build</li> <li>Integrated application suite on a shared data layer</li> <li>Single server-side development language, CH</li> <li>Alternative content format, e.g. ISON</li> <li>Enterprise licences</li> <li>MindFusion upgrade</li> <li>Standards-based API</li> <li>Compatibility with NHSD framework</li> <li>Future proof (Cloud, standard API, microservices, containerisation strategy)</li> <li>Leverages NHSD infrastructure security layer</li> </ul>
ironment. Still, it has the capability to migrate to more fle	HTTP, a single database layer, and thin internal API for authoring sible and advance architectures if required. The current legacy monolith services to increase scalability, performance, and manageability. Sitate more flexible methods of consuming it.

Note: Infrastructure requirement for content creation and delivery to Supplier is significantly different from those to deliver the UEC Triage service.

## Cloud First

сом	том
<ul> <li>Hosted on servers</li> <li>Desktop application</li> <li>Some web applications</li> <li>Some API</li> <li>The data centre is being shut down</li> </ul>	<ul> <li>Cloud service using data layer capabilities</li> <li>Elastic performance to cope with remote consumer users</li> <li>Remote access to content for training, and cross-reference</li> <li>High availability</li> <li>NHSD build specification</li> <li>NHSD Class 1 protection (no patient data)</li> </ul>

All services in the hosting datacentre are being moved (forklift) to cloud. This is an opportunity to build a cloud-first service based on the existing capabilities, with an enhanced data model. The new applications will have the same capability but will be rewritten in a common development language using Test Driven Development. This will support future Agile development of pathways tools and capabilities.

#### Standard API

сом	том
<ul> <li>Proprietary API between tools</li> <li>Some web applications</li> <li>Proprietary integration with DoS</li> </ul>	<ul> <li>Standard API for import/export assertions (inc. national services)</li> <li>Thin API between authoring apps for performance</li> <li>Standards API to external service (JIRA, Teams, SharePoint)</li> <li>Future proof for access to the content by other services</li> <li>A standards-based application security layer</li> <li>JWT user authentication method</li> </ul>

Moving to standard API inline with NHSD specification will enable the author to access other external services during content development. It will also enable external application and service to access the published content is a more flexible way; for example, 111-online, PaCCS CDS, and other services that Pathways may develop.

#### Future capability - Authoring process

сом	TOM
A complex process, difficult to learn     Bespoke tools and processes     Difficult to model and test     Fragmented tools and processes	Linear pathway development for a given condition Known start and endpoints Preload the pathway with demographics and assertions Selection of assertions Concerns/condition to be validated or excluded Auto navigates through a sequence of Q/A as waypoints to give the best route to a disposition across the exiting data set. Authoring support, outcome route mapping/prediction Integrated document management Integrated workflow management Visualisation and development environment Compactable with SNOMED CT

#### Enhancing and exploiting an existing mature data set.

The proposed solution builds the foundation for future agile development of content and services. Intelligent process and support tools will enable differing authoring methodologies across the same data set and provide greater secondary use of the Pathways content. More straightforward tools and processes will allow a broader range of authoring contribution and peer review and will enable collaborative authoring with external specialist resources. The use of registry will enable the content to (and application) to be linked to trusted content and multimedia. All and rules-based assisted linear pathway creation will make use of existing content and accelerate the creation of more target and specialised pathway. Bi and other supporting information will improve quality and safety during the authoring process.

### 3.3. Initial Statement/s of Work (SOW/s)

### 3.3.1. SOW 1

SOW 1 is for the Supplier to deliver the BAU service requirements. NHS Pathways can assist in the key challenges facing the Urgent and Emergency Care landscape by delivering on specific outcomes. Delivery of the high-level requirements above by a DevOps team will significantly assist in meeting those outcomes.

### 3.3.2. SOW 2

SoW 2 is for the supplier to deliver a 6-week piece of work that will put together the knowledge gathering, high level overall plan, detailed plan and some technical details for a subsequent SoW with a longer timeframe and larger team.

Further SOWs will be raised for the future Re-platforming phases as the previous phases approach completion and for future periods of the BAU service requirements.

## 4. Key Milestones and Call-Off Deliverables

The individual SOWs contain the key milestones and deliverables information.

Key Deliverable	Milestone/	Description	Timeframe or Delivery Date
		Left blank	Left Blank

## 5. Responsibilities of the Parties

Both Buyer and Supplier have key roles in and are responsible (individually and crucially in partnership) for the successful delivery of the outcomes in this call-off contract, and the Statements of Work related to it.

High-level information on how the supplier is expected to work and the key roles in NHS Digital that it is anticipated they will be working with is provided in section 2.7.

## 6. Skills / Capabilities Profile

Although rates will be commercially evaluated on the basis of a sample profile, the listing below is intended to provide the Supplier with an initial idea. It is not intended, at this level, to be definitive (individual Statements of Work should be more specific in this regard).

Resource Profile details are contained in:	Annex 5

# Annex 1: Call-Off Schedule 20 – Target Sectors

HEALTH AND SOCIAL CARE SETTINGS	Tick boxes
Primary Care	N
Ambulance Services	Y
Secondary Care - Hospital Settings	N
Pharmacy	N
Mental Health	N
Community Care - Childrens Services	N
Community Care - Adult Services	N
Public Health & Wellbeing	N
Screening	N
Social Care - Childrens Services	N
Social Care - Adult Services	N
Genomics	N
Health and Social Care Policy	N
Health Informatics	N
Medicines and Healthcare Products	N
Health and Social Care Regulation / Quality	N
Health Sector Education, Training and Workforce	N
Health and Social Care Research	N
Blood and Transplant Services	N
Independent Health Provision	N

ſ	Urgent Emergency Care	Y
ь.		

HEALTH AND SOCIAL CARE SYSTEMS	
SPINE (Summary Care Record)	N
Screening Systems	N
Electronic Prescription Service (EPS)	N
Electronic Referral Service (ERS)	N
GP IT Systems & Services	N
Health and Social Care Mobile Apps	N
Health and Social Care Web Apps	N
Citizen Identification and Verification Services	N
Health System Infrastructure (email, etc)	N
Secondary Uses Services	N
Health Data Collection, Processing and Dissemination	N
Care Management Systems	N

# Annex 2: Call-Off Schedule 20 – Technology Capabilities

Application Development	Tick boxes
Continuous Integration & Delivery Tools	Y
Testing & Quality Assurance Tools	Y

iness Applications	Tick boxes
Data Warehousing	N
Enterprise Applications	N
Geospatial	N
Project Management	Y

Customer Management	Tick boxes
CRM	
Enterprise Applications	N

Management	Tick boxes
Middleware	N
Networking	N
Service Management	N
System Management	N

Services	Tick boxes
Anti-Virus, Vulnerability Mgt & Monitoring	N
Cloud Orchestration	N
Encryption	N
Remote Access Service	N

(continued on next page)

oftware Infrastructure	Tick boxes
Enterprise Architecture Tools	
Architecture Tools	Y
Intelligent Business Process Management Suites	
Business Process Management	Y
Architecture Tools	Y
Discovery / Search	Y
Frameworks, Languages, & Libraries	Y
Identity & Access Management	N
Non-Relational Databases	Y
Performance & Availability Monitoring	N
Relational Databases	Y
Server Technology	N
Server/Desktop OS	N
Serverless	Y
Source Code Management	Y
Storage	Y
Virtualisation & Containerisation	Y
Visualisation Tools	N
Web Analytics	N

# Annex 3: Call-Off Schedule 20 – EPIC Template

Template not applicable as referenced in 2.8 Business Outcomes

The Buyer uses epics, stories, and tasks recorded in Jira to describe, track, manage and assure the work it is undertaking.

For BAU, the Supplier is required to deliver a range of services to ensure the effective development, operation and continuous improvement of the live service as detailed in SOW 1.

For re-platforming, the Supplier is required to deliver a range of services to meet the requirements identified in the individual SOWs that will be raised to complete each Phase of the re-platforming journey, enabling the overarching business outcomes described earlier to be achieved.

# Annex 4: Call-Off Schedule 20 – Product Backlog Item List

Table not applicable as referenced in 2.8 Business Outcomes

For the Pathways re-platforming element, the high-level themes to meet the vision and address the key issues, as outlined in section 2.8, are-

- Simplified authoring (clinical content)
- Simplified reporting (data)
- SNOMED coded (standardised) inputs and outputs
- Simplified distribution of content (releases)
- · Easier updating of content
- Simplified outcome mapping to services available (linking to Directory of Services)

It is anticipated that the following areas will be covered by this call-off contract and will form the basis of the work required to deliver the backlog that will be developed, in conjunction with the Supplier, to achieve these 6 themes. This list is not exhaustive:

- · Supplier transition and handover activities.
- · Agile delivery planning and management
- Discovery and knowledge gathering
- User research and design
- Data modelling
- Business Analysis
- Scoping and developing Alphas
- Scoping and developing Betas
- Integration and interoperability
- Specialist clinical skills e.g.: clinical terms modelling and clinical safety
- · Functional and non-functional testing
- Transition to live operation

This backlog will be continuously refined by the Buyer, with support from the Supplier, to reflect the evolving work. In the context of this call-off contract, the Supplier is responsible for the operational management of the sprint planning process, e.g. sprint planning, user stories etc.

The Supplier will also deliver solution assurance activities during the period of the call-off contract including ensuring sign-off by the Buyer.

# Annex 5: Call-Off Schedule 20 – Resource Profile

Although rates will be commercially evaluated based on a sample profile, the listing below is intended to provide the Supplier with an initial idea. It is not intended, at this level, to be definitive (individual Statements of Work should be more specific in this regard).

This indicative list combines both the BAU service and the Re-platforming resource estimates, noting that the resources for the BAU service will be for the duration of the call off whereas resources for Re-platforming will be for the first 1 to 2 years of the call off only:-

Data ScientistPerformance Analystpsp	DDaT Cluster	Role Family
Performance AnalystpsBusiness Relationship ManagerpsChange and Release ManagerpsCommand and ControlpsApplications OperationspsEngineer End UserpsEngineer InfrastructurepsIncident ManagerpsProblem ManagerpsService Desk ManagerpsService Transition ManagerpsDeliverypsDeliverypsDeliverypsTest EngineernnicalData ArchitectnnicalNetwork ArchitectnnicalSoftware DevelopernnicalSoftware DevelopernnicalTechnical Architect	Data	Data Engineer
psBusiness Relationship ManagerpsChange and Release ManagerpsCommand and ControlpsApplications OperationspsEngineer End UserpsEngineer InfrastructurepsIncident ManagerpsProblem ManagerpsService Desk ManagerpsService Transition ManagerpsBusiness AnalysispsDeliveryProduct ManagerpsTest EngineermicalData ArchitectpnicalInfrastructure EngineermicalNetwork ArchitectmicalSoftware DevelopermicalTechnical ArchitectmicalTechnical ArchitectmicalTechnical ArchitectmicalTechnical ArchitectmicalTechnical ArchitectmicalTechnical ArchitectmicalTechnical ArchitectmicalTechnical Architect	Data	Data Scientist
psChange and Release ManagerpsCommand and ControlpsApplications OperationspsEngineer End UserpsEngineer InfrastructurepsIncident ManagerpsIT Service ManagerpsProblem ManagerpsService Desk ManagerpsService Transition ManagerpsBusiness Analysisbuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryDeliverybuct DeliveryProduct Managerbuild DeliveryDeliverybuct DeliveryProduct Managerbuild DeliveryProduct Managerbuild DeliveryProduct Managerbuild DeliveryProduct Managerbuild DeliveryProduct Managerbuild DevOpsInfrastructure Engineerbuild DevOpsInfrastructure Engineerbuild DevOpsSecurity Architectbuild DevOpsSecurity Architect	Data	Performance Analyst
psCommand and ControlpsApplications OperationspsEngineer End UserpsEngineer InfrastructurepsIncident ManagerpsIT Service ManagerpsProblem ManagerpsService Desk ManagerpsService Transition ManagerpsDeliverybuct DeliveryProduct ManagerpsCommand and Controlbuct DeliveryProblem Managerbuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryDeliveryProduct ManagerData ArchitectbuncalInfrastructure EngineerbuncalNetwork ArchitectbuncalSoftware DeveloperbuncalTechnical Architect	IT Ops	Business Relationship Manager
psApplications OperationspsEngineer End UserpsEngineer InfrastructurepsIncident ManagerpsIT Service ManagerpsProblem ManagerpsService Desk ManagerpsService Transition ManagerpsDeliverybuct DeliveryDeliverybuct DeliveryProduct ManagerpsCAT AnalystTest EngineerTest EngineerTest ManagerData ArchitectpnicalInfrastructure EngineernnicalNetwork ArchitectnnicalSecurity ArchitectnnicalSecurity ArchitectnnicalTechnical ArchitectnnicalTechnical ArchitectnnicalTechnical Architect	IT Ops	Change and Release Manager
psEngineer End UserpsEngineer InfrastructurepsIncident ManagerpsIT Service ManagerpsProblem ManagerpsService Desk ManagerpsService Transition Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryProduct ManagerburgerData ArchitectburgerData ArchitectburgerDevOpsburgerNetwork ArchitectburgerSecurity ArchitectburgerSoftware DeveloperburgerTechnical Architect	IT Ops	Command and Control
psEngineer InfrastructurepsIncident ManagerpsIT Service ManagerpsProblem ManagerpsService Desk ManagerpsService Transition Managerbuct DeliveryBusiness Analysisbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryDeliverybuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryProduct ManagerburgerDeliveryburgerDeliveryburgerDeliveryburgerDeliveryburgerDevopsburgerDevOpsburgerInfrastructure EngineerburgerSecurity ArchitectburgerSoftware DeveloperburgerTechnical Architect	IT Ops	Applications Operations
psIncident ManagerpsIT Service ManagerpsProblem ManagerpsService Desk ManagerpsService Transition Managerbuct DeliveryBusiness Analysisbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryProduct Managerbutt DeliveryDevopsbutt DeliverySecurity Architectbutt DeliverySoftware Developerbutt DeliveryTechnical Architect	IT Ops	Engineer End User
psIT Service ManagerpsProblem ManagerpsService Desk ManagerpsService Transition Managerbuct DeliveryBusiness Analysisbuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryProduct Managerbusiness AnalystTest Engineerbusiness AnalystDevOpsbusinessInfrastructure EngineerbusinessNetwork ArchitectbusinessSoftware DeveloperbusinessTechnical Architect	IT Ops	Engineer Infrastructure
psProblem ManagerpsService Desk ManagerpsService Transition Managerbuct DeliveryBusiness Analysisbuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryDevOpsbuncalData ArchitectbuncalInfrastructure EngineerbuncalNetwork ArchitectbuncalSoftware DeveloperbuncalTechnical Architect	IT Ops	Incident Manager
psProblem ManagerpsService Desk ManagerpsService Transition Managerbuct DeliveryBusiness Analysisbuct DeliveryProduct Managerbuct DeliveryDeliverybuct DeliveryProduct Managerbuct DeliveryProduct Managerbuct DeliveryDevOpsbuncalData ArchitectbuncalInfrastructure EngineerbuncalNetwork ArchitectbuncalSoftware DeveloperbuncalTechnical Architect	IT Ops	IT Service Manager
psService Transition Managerluct DeliveryBusiness Analysisluct DeliveryDeliveryluct DeliveryProduct ManagerQAT AnalystTest EngineerTest EngineerTest ManagerData ArchitectDevOpsInnicalInfrastructure EngineerInnicalSecurity ArchitectSecurity ArchitectSoftware DeveloperTechnical ArchitectTechnical Architect	IT Ops	The share and the second strength and the share and the strength and the strength and the second strength and the strength an
psService Transition Managerluct DeliveryBusiness Analysisluct DeliveryDeliveryluct DeliveryProduct ManagerQAT AnalystTest EngineerTest EngineerTest ManagerData ArchitectDevOpsInnicalInfrastructure EngineerInnicalSecurity ArchitectSecurity ArchitectSoftware DeveloperTechnical ArchitectTechnical Architect	IT Ops	Service Desk Manager
Juct Delivery     Delivery       Juct Delivery     Product Manager       QAT Analyst     Test Engineer       Test Manager     Data Architect       Delivery     Data Architect       Innical     DevOps       Infrastructure Engineer     Network Architect       Security Architect     Security Architect       Innical     Devops       Infrastructure Engineer     Test Manager	IT Ops	Service Transition Manager
Iuct Delivery       Product Manager         QAT Analyst       Test Engineer         Test Manager       Data Architect         Innical       DevOps         Innical       Infrastructure Engineer         Network Architect       Security Architect         Innical       Software Developer         Inical       Technical Architect	Product Delivery	Business Analysis
QAT Analyst       Test Engineer       Test Manager       Data Architect       DevOps       Infrastructure Engineer       Network Architect       Security Architect       Software Developer       Technical Architect	Product Delivery	Delivery
Test Engineer Test Manager Data Architect DevOps Infrastructure Engineer Network Architect Security Architect Software Developer Technical Architect	Product Delivery	Product Manager
Test Manager Data Architect DevOps Infrastructure Engineer Network Architect Security Architect Software Developer Technical Architect	QAT	QAT Analyst
Test ManagernnicalData ArchitectnnicalDevOpsnnicalInfrastructure EngineernnicalNetwork ArchitectnnicalSecurity ArchitectnnicalSoftware DevelopernnicalTechnical Architect	QAT	Test Engineer
DevOps           Infrastructure Engineer           Inical           Inical           Inical           Inical           Inical           Security Architect           Software Developer           Technical Architect	QAT	
Infrastructure Engineer       Inical       Inical       Inical       Security Architect       Inical       Software Developer       Inical       Technical Architect	Technical	Data Architect
Innical         Network Architect           Innical         Security Architect           Innical         Software Developer           Innical         Technical Architect	Technical	DevOps
nnical Network Architect Security Architect Software Developer Technical Architect	Technical	Infrastructure Engineer
nnical Software Developer Technical Architect	Technical	
nnical Software Developer Technical Architect	Technical	Security Architect
No 1964 PA	Technical	
Centred Design Content Designer	Technical	Technical Architect
	User Centred Design	Content Designer
Centred Design Graphic Interaction Designer	User Centred Design	Graphic Interaction Designer

C52340 NHS Pathways 03 Call Off Schedules 1.0F 22Dec21

Call-Off Schedule 20 (Call-Off Specification) Call-Off Ref: Crown Copyright 2018

User Centred Design	Service Designer	
User Centred Design	Technical Writer	
User Centred Design	User Researcher	
No DDaT Cluster Mapping	Cyber Security	

Framework Ref: RM6221 Project Version: v1.0 Model Version: v3.0